UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the Quarterly Period Ended March 31, 2021

Commission File Number 001-35570

SONNET BIOTHERAPEUTICS HOLDINGS, INC.

(Exact name of registrant as specified in the charter)

Delaware		20-2932652
(State or other jurisdiction of		(I.R.S. Employer
incorporation or organization)		Identification Number)
	erlook Center, Suite 102, Princeton, NJ eess of principal executive offices) (Zip Co	
Registrant's te	elephone number, including area code: (60)	9) 375-2227
Securities registered pursuant to Section 12(b) of the Act:		
Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common Stock, \$0.0001 Par Value	SONN	The Nasdaq Stock Market LLC
Indicate by check mark whether the registrant (1) has filed all reports months (or for such shorter period that the registrant was required to		
Indicate by check mark whether the registrant has submitted elect (§232.405 of this chapter) during the preceding 12 months (or for such		
Indicate by check mark whether the registrant is a large accelerate company. See the definitions of "large accelerated filer," "accelerated		
	ge accelerated filer [] Accelerated filer [elerated filer [X] Smaller reporting compa Emerging growth company []	
If an emerging growth company, indicate by check mark if registra accounting standards pursuant to Section 13(a) of the Exchange Act.		ansition period for complying with any new or revised financial
Indicate by check mark whether the registrant is a shell company (as	defined in Rule 12b-2 of the Exchange Ad	ct). [] Yes [X] No.
Indicate the number of shares outstanding of each of the registrant's par value \$0.0001 of Sonnet BioTherapeutics Holdings, Inc. issued a		practicable date. There were 21,197,290 shares of common stock,
Sonnet BioTherapeutics Holdings, Inc. and Subsidiaries		
INDEV		

INDEX

		Page No.
Part I	Financial Information	3
Item 1:	Financial Statements (unaudited)	3
	Consolidated Balance Sheets	3
	Consolidated Statements of Operations	4
	Consolidated Statements of Changes in Stockholders' Equity (Deficit)	5
	Consolidated Statements of Cash Flows	6
	Notes to Interim Consolidated Financial Statements	7
Item 2:	Management's Discussion and Analysis of Financial Condition and Results of Operations	15
Item 3:	Quantitative and Qualitative Disclosures about Market Risk	28
Item 4:	Controls and Procedures	28
Part II	Other Information	29

Item 1:	<u>Legal Proceedings</u>	29
Item 1A:	Risk Factors	29
Item 2:	Unregistered Sales of Equity Securities and Use of Proceeds	30
Item 3:	Defaults Upon Senior Securities	30
Item 4:	Mine Safety Disclosures	30
Item 5:	Other Information	31
Item 6:	Exhibits	31
Signatures		32

2

PART I

ITEM 1: FINANCIAL STATEMENTS

Sonnet BioTherapeutics Holdings, Inc. Consolidated Balance Sheets (unaudited)

	I	March 31, 2021	Se	eptember 30, 2020
Assets			_	
Current assets:				
Cash	\$	6,738,811	\$	7,349,903
Prepaid expenses and other current assets		659,202		287,738
Total current assets		7,398,013		7,637,641
Property and equipment, net		61,726		67,889
Operating lease right-of-use asset		165,773		205,919
Other assets				82,959
Total assets	\$	7,625,512	\$	7,994,408
Liabilities and stockholders' equity				
Current liabilities:				
Related-party notes	\$	748	\$	21,184
Accounts payable		1,913,532		2,057,559
Accrued expenses		2,895,736		2,063,678
Operating lease liability		88,083		82,060
Deferred income		500,000		500,000
Total current liabilities		5,398,099		4,724,481
Note payable		125,501		124,878
Operating lease liability		79,603		125,132
Total liabilities		5,603,203		4,974,491
Commitments and contingencies (Note 6)				
Stockholders' equity:				
Preferred stock; \$0.0001 par value: 5,000,000 shares authorized. No shares issued or outstanding				
Common stock; \$0.0001 par value: 125,000,000 shares authorized; 21,197,290 and 14,724,105 issued and		_		_
outstanding at March 31, 2021 and September 30, 2020, respectively		2,119		1.472
Additional paid-in capital		50,641,794		39,723,702
Accumulated deficit		(48,621,604)		(36,705,257)
Total stockholders' equity		2,022,309		3,019,917
Total liabilities and stockholders' equity	\$	7,625,512	\$	7,994,408

See accompanying notes to unaudited interim consolidated financial statements

3

Sonnet BioTherapeutics Holdings, Inc. Consolidated Statements of Operations (unaudited)

	Three Months Ended March 31,				 Six Months En	March 30,	
		2021		2020	2021		2020
Operating expenses:					 		
Research and development	\$	3,840,399	\$	1,302,515	\$ 7,706,407	\$	2,710,663
General and administrative		2,196,632		1,208,374	4,194,617		2,269,280
Loss from operations		(6,037,031)		(2,510,889)	(11,901,024)		(4,979,943)
Foreign exchange gain (loss)		(2,076)		14,142	 (15,323)		14,142
Net loss		(6,039,107)		(2,496,747)	(11,916,347)		(4,965,801)
Share information:					 		
Net loss per share, basic and diluted	\$	(0.32)	\$	(0.44)	\$ (0.66)	\$	(0.88)
Weighted average shares outstanding, basic and diluted		18,742,926		5,719,988	17,972,329		5,655,591

See accompanying notes to unaudited interim consolidated financial statements

Sonnet BioTherapeutics Holdings, Inc. Consolidated Statements of Changes in Stockholders' Equity (Deficit) (unaudited)

	Commo	on stock	r	Ad	ditional paid- in		Accumulated	
	Shares		Amount		capital	1	deficit	Total
Balance at October 1, 2020	14,724,105	\$	1,472	\$	39,723,702	\$	(36,705,257)	\$ 3,019,917
Warrant exercises	23,863		2		_		_	2
Net share settlement of warrants	2,427,761		243		(243)		_	_
Share-based compensation	_		_		370,055		_	370,055
Net loss	_		_		_		(5,877,240)	(5,877,240)
Balance at December 31, 2020	17,175,729		1,717		40,093,514		(42,582,497)	(2,487,266)
Sale of common stock, net of issuance costs	4,021,561		402		10,178,225		_	10,178,627
Share-based compensation	_		_		370,055		_	370,055
Net loss	_		_		_		(6,039,107)	(6,039,107)
Balance at March 31, 2021	21,197,290	\$	2,119	\$	50,641,794	\$	(48,621,604)	\$ 2,022,309
				Ad	ditional paid-			
	Commo	on stock			in	I	Accumulated	
	Shares		Amount		capital		deficit	Total
Balance at October 1, 2019	5,547,643	\$	555	\$	9,594,100	\$	(12,440,142)	\$ (2,845,487)
Sale of common stock, net of issuance costs	128,313		13		2,715,017			2,715,030
Issuance of common stock to settle related-party notes	8,526		1		199,999		_	200,000
Net loss	_		_		_		(2,469,054)	(2,469,054)
Balance at December 31, 2019	5,684,482		569		12,509,116		(14,909,196)	(2,399,511)
Sale of common stock, net of issuance costs	57,762		5		1,354,995			1,355,000
Net loss	_		_		_		(2,496,747)	(2,496,747)
Balance at March 31, 2020	5,742,244	\$	574	\$	13,864,111	\$	(17,405,943)	\$ (3,541,258)

See accompanying notes to unaudited interim consolidated financial statements

- 4

Sonnet BioTherapeutics Holdings, Inc. Consolidated Statements Cash Flows (unaudited)

		Six Months Ended March 31,			
		2021	2020		
Cash flows from operating activities:					
Net loss	\$	(11,916,347)	\$	(4,965,801)	
Adjustments to reconcile net loss to net cash used in operating activities:					
Depreciation		6,163		2,131	
Amortization of operating lease right-of-use asset		40,146		12,205	
Share-based compensation		740,110			
Non-cash interest		624		(14,142)	
Change in operating assets and liabilities:					
Prepaid expenses and other current assets		(371,464)		(15,383)	
Other assets		82,959		(82,959)	
Accounts payable		(144,027)		1,835,322	
Accrued expenses		832,058		(310,590)	
Operating lease liability		(39,507)		(11,885)	
Net cash used in operating activities		(10,769,285)		(3,551,102)	
Cash flows from investing activities:					
Purchases of property and equipment		_		(65,265)	
Net cash used in investing activities		_		(65,265)	
Cash flows from financing activities:					
Proceeds from the issuance of common stock, net of issuance costs		10,178,627		4,070,030	
Related party receivable		_		(200,000)	
Proceeds from the exercise of warrants		2			
Proceeds received from related-party notes		_		30,000	
Repayments of related-party notes		(20,436)		(46,461)	
Net cash provided by financing activities		10,158,193		3,853,569	
1 , ,					
Net (decrease) increase in cash		(611,092)		237,202	
Cash, beginning of period		7,349,903		35,653	
Cash, end of period	\$	6,738,811	\$	272,855	
Supplemental disclosure of non-cash investing and financing activities:					
Net settlement of warrants	\$	243	\$	_	
Issuance of common stock to settle related-party notes	\$	_	\$	200,000	
Right of use asset and liability recorded upon adoption of ASC 842	\$		\$	255,938	
6	<u>\$</u>		Ψ	255,756	

See accompanying notes to unaudited interim consolidated financial statements

Sonnet BioTherapeutics Holdings, Inc. Notes to Unaudited Interim Consolidated Financial Statements

1. Organization and description of business

Description of business

Sonnet BioTherapeutics, Inc. ("Sonnet") was incorporated as a New Jersey corporation on April 6, 2015. Sonnet is a clinical stage, oncology-focused biotechnology company with a proprietary platform for innovating biologic medicines of single- or bi-specific action. Known as F_HAB™ (Fully Human Albumin Binding), the technology utilizes a fully human single chain antibody fragment (scFv) that binds to and "hitch-hikes" on human serum albumin (HSA) for transport to target tissues. Sonnet's lead proprietary asset, SON-1010, is a fully human version of Interleukin 12 (IL-12), covalently linked to the F_HAB construct, for which Sonnet intends to pursue clinical development in solid tumor indications, including non-small cell lung cancer and head and neck cancer. Sonnet has completed a nonhuman primate (NHP) GLP toxicity study with SON-1010 and is preparing an Investigational New Drug (IND) application for submission to the FDA with the goal of initiating a Phase 1 clinical trial during the second half of 2021. The Company acquired the global development rights to its most advanced compound, a fully human version of Interleukin 6 (IL-6), in April 2020. Sonnet is advancing its SON-080 candidate in target indications of Chemotherapy-Induced Peripheral Neuropathy (CIPN) and Diabetic Peripheral Neuropathy (DPN). Sonnet intends to file an IND for a US Phase 1b/2a pilot-scale efficacy study with SON-080 in CIPN during the second half of 2021. Pursuant to a license agreement the Company entered with New Life in May 2021, New Life will be responsible for leading the development program for SON-080 in DPN with the objective of initiating an ex-US Phase 1b/2a pilot-scale efficacy study during the second half of 2021.

On April 1, 2020, Sonnet completed its merger (the "Merger") with publicly-held Chanticleer Holdings, Inc. ("Chanticleer") in accordance with the terms of the Plan of Merger dated October 10, 2019, as amended by Amendment No. 1 on February 7, 2020 (the "Merger Agreement"). Immediately prior to the Merger, Chanticleer spun-off its restaurant operations to a spin-off entity and no assets or liabilities of the restaurant business remained after the spin-off. After the Merger, Chanticleer changed its name to Sonnet BioTherapeutics Holdings, Inc. ("Sonnet Holdings" or the "Company") and is focused on advancing Sonnet's pipeline of oncology candidates and the strategic expansion of Sonnet's technology platform into other human disease.

Global pandemic - COVID-19

On March 10, 2020, the World Health Organization characterized the novel COVID-19 virus as a global pandemic. There is significant uncertainty as to the likely effects of this disease which may, among other things, materially impact the Company's planned clinical trials. This pandemic or outbreak could result in difficulty securing clinical trial site locations, clinical research organizations ("CROs"), and/or trial monitors and other critical vendors and consultants supporting the trial. In addition, outbreaks or the perception of an outbreak near a clinical trial site location could impact the Company's ability to enroll patients. These situations, or others associated with COVID-19, could cause delays in the Company's clinical trial plans and could increase expected costs, all of which could have a material adverse effect on the Company's business and its financial condition. At the current time, the Company is unable to quantify the potential effects of this pandemic on its future operations.

Liquidity

The Company has incurred recurring losses and negative cash flows from operations activities since inception and it expects to generate losses from operations for the foreseeable future primarily due to research and development costs for its potential product candidates. The Company believes its cash of \$6.7 million at March 31, 2021 will fund the Company's projected operations into July 2021. Substantial additional financing will be needed by the Company to fund its operations. These factors raise substantial doubt about the Company's ability to continue as a going concern. The accompanying consolidated financial statements have been prepared on a going concern basis, which contemplates the realization of assets and satisfaction of liabilities in the normal course of business. The consolidated financial statements do not include any adjustments that might result from the outcome of this uncertainty.

7

Sonnet BioTherapeutics Holdings, Inc. Notes to Unaudited Interim Consolidated Financial Statements

The Company entered into an At-the-Market Sales Agreement with BTIG, LLC ("BTIG") on February 5, 2021 (the "Sales Agreement"). Pursuant to the Sales Agreement, the Company may offer and sell, from time to time, through BTIG, as sales agent and/or principal, shares of its common stock, having an aggregate offering price of up to \$15,875,000, subject to certain limitations on the amount of common stock that may be offered and sold by the Company set forth in the Sales Agreement. The Company is not obligated to make any sales of shares under the Sales Agreement and any determination by the Company to do so will be dependent, among other things, on market conditions and the Company's capital raising needs. Through March 31, 2021 the Company sold an aggregate of 4,021,561 shares under the Sales Agreement for gross proceeds of \$10.6 million and net proceeds of \$10.2 million.

On May 2, 2021, the Company entered into a License Agreement (the "Agreement") with New Life Therapeutics PTE, LTD. (See Note 9).

The Company plans to secure additional capital in the future through equity or debt financings, partnerships, collaborations, or other sources to carry out the Company's planned development activities. If additional capital is not available when required, the Company may need to delay or curtail its operations until such funding is received. Various internal and external factors will affect whether and when the Company's product candidates become approved for marketing and successful commercialization. The regulatory approval and market acceptance of the Company's products candidates, length of time and cost of developing and commercializing these product candidates and/or failure of them at any stage of the approval process will materially affect the Company's financial condition and future operations.

Operations since inception have consisted primarily of organizing the Company, securing financing, developing its technologies through performing research and development and conducting preclinical studies. The Company faces risks associated with companies whose products are in development. These risks include the need for additional financing to complete its research and development, achieving its research and development objectives, defending its intellectual property rights, recruiting and retaining skilled personnel, and dependence on key members of management.

2. Summary of Significant Accounting Policies

a. Basis of presentation

The accompanying unaudited interim consolidated financial statements have been prepared in conformity with U.S. generally accepted accounting principles ("U.S. GAAP") for interim financial information as found in the Accounting Standard Codification ("ASC") and Accounting Standards Updates (ASUs") of the Financial Accounting Standards Board ("FASB"). In the opinion of management, the accompanying unaudited interim consolidated financial statements include all normal and recurring adjustments (which consist primarily of accruals, estimates and assumptions that impact the unaudited interim financial statements) considered necessary to present fairly the Company's financial position as of March 31, 2021 and its results of operations and cash flows for the three and six months ended March 31, 2021 and 2020. The unaudited interim consolidated financial statements presented herein do not contain the required disclosures under U.S. GAAP for annual financial statements and should be read in conjunction with the annual audited financial statements and related notes of Sonnet Holdings as of and for the year ended September 30, 2020 included in the Company's Annual Report on Form 10-K for the fiscal year ended September 30, 2020.

b. Consolidation

The consolidated financial statements include the accounts of the Company and its wholly-owned subsidiaries. All intercompany accounts and transactions have been eliminated in consolidation.

0

Sonnet BioTherapeutics Holdings, Inc. Notes to Unaudited Interim Consolidated Financial Statements

c. Use of estimates

The preparation of consolidated financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities as of the date of the consolidated financial statements and the reported amounts of expenses during the reporting period. Actual results could differ from those estimates. Estimates and assumptions are periodically reviewed, and the effects of revisions are reflected in the consolidated financial statements in the period they are determined to be necessary. Significant estimates include the recording of prepayments and accruals related to research and development.

d. Property and equipment

Property and equipment are recorded at cost and depreciated using the straight-line method over the estimated useful lives of the assets. Expenditures for repairs and maintenance that do not extend the estimated useful life or improve an asset are expensed as incurred. Upon retirement or sale, the cost and related accumulated depreciation and amortization of assets disposed of are removed from the accounts, and any resulting gain or loss is included in the statement of operations. As of March 31, 2021, the property and equipment balance was comprised of leasehold improvements and computer equipment associated with the Princeton office lease discussed in Note 5.

e. Net loss per share

Basic net loss per share is computed by dividing net loss by the weighted-average number of shares of common stock outstanding during each period (and potential shares of common stock that are exercisable for little or no consideration). Included in basic weighted-average number of shares of common stock outstanding during the three and six months ended March 31, 2021 are the Series B warrants with an exercise price of \$0.0001 per share.

Diluted loss per share includes the effect, if any, from the potential exercise or conversion of securities such as common stock warrants and stock options which would result in the issuance of incremental shares of common stock. For diluted net loss per share, the weighted-average number of shares of common stock is the same for basic net loss per share due to the fact that when a net loss exists, dilutive securities are not included in the calculation as the impact is anti-dilutive.

The following potentially dilutive securities have been excluded from the computation of diluted weighted-average shares of common stock outstanding as they would be anti-dilutive:

	March 31, 2021	March 31, 2020
Warrants	105,812	105,812
Legacy Chanticleer warrants	17,760	_
Series C warrants	11,329,463	_
Unvested restricted stock	653,845	
	12,106,880	105,812

Sonnet BioTherapeutics Holdings, Inc. Notes to Unaudited Interim Consolidated Financial Statements

f. Recent accounting pronouncements

Recently Announced

In December 2019, the FASB issued ASU 2019-12, "Income Taxes Topic 740-Simplifying the Accounting for Income Taxes" ("ASU 2019-12"), which intended to simplify various aspects related to accounting for income taxes. ASU 2019-12 removes certain exceptions to the general principles in Topic 740 and also clarifies and amends existing guidance to improve consistent application of Topic 740. This guidance is effective for fiscal years beginning after December 15, 2020, including interim periods therein, and early adoption is permitted. The Company is currently evaluating the new standard to determine the potential impact on its financial condition, results of operations, cash flows, and financial statement disclosures.

Recently Adopted

In August 2018, the FASB issued ASU 2018-13, Disclosure Framework- Changes to the Disclosure Requirements for Fair Value Measurements, which changes the fair value measurement disclosure requirements of ASC 820. The goal of the ASU is to improve the effectiveness of ASC 820's disclosure requirements. The adoption of ASU 2018-13 on October 1, 2019, did not have a material impact on the consolidated financial statements.

3. Accrued Expenses

Accrued expenses consisted of the following:

N	larch 31, 2021	Se	eptember 30, 2020
\$	1,474,167	\$	1,065,398
	1,204,792		519,159
	215,187		479,121
	1,590		
\$	2,895,736	\$	2,063,678
	\$ \$	\$ 1,474,167 1,204,792 215,187 1,590	\$ 1,474,167 \$ 1,204,792 \$ 215,187 \$ 1,590

During the six months ended March 31, 2020, the Company issued unsecured notes payable to various related parties resulting in cash proceeds of \$30,000. These notes are payable on demand and payments of \$20,436 and \$46,461 were made during the six months ended March 31, 2021 and 2020, respectively. The interest on these notes was de minimis during each of those periods.

In October 2019, the Company issued 8,526 shares of common stock to settle \$0.2 million of related party notes.

1.0

Sonnet BioTherapeutics Holdings, Inc. Notes to Unaudited Interim Consolidated Financial Statements

PPP Loan

On March 27, 2020, the U.S. federal government enacted the Coronavirus Aid, Relief, and Economic Security Act ("CARES Act"). The CARES Act includes a provision for a Paycheck Protection Program ("PPP"), administered by the U.S. Small Business Administration ("SBA") and further amended by the Paycheck Protection Program Flexibility Act of 2020 ("PPP Flexibility Act"), which was enacted on June 5, 2020.

In May 2020, the Company received a PPP Loan of \$0.1 million. The application for these funds required the Company to certify in good faith that current economic uncertainty made the loan request necessary to support the ongoing operations of the Company. The Company was also required to certify that the loan funds would be used to retain workers and maintain payroll or make mortgage payments, lease payments, and utility payments. The PPP Loan has a two-year term and bears interest at a rate of 1.0% per year.

Under the terms of the CARES Act, the Company can apply for and be granted forgiveness for all or a portion of the PPP Loan. Such forgiveness, if any, will be determined, subject to limitations, based on the use of loan proceeds for payroll costs, rent and utility costs and provided that only a portion of the use of proceeds are for non-payroll costs. The unforgiven portion of the PPP Loan may be repaid by the Company at any time prior to maturity with no prepayment penalty. While the Company believes that its use of the loan proceeds will meet the conditions for forgiveness of the PPP Loan, at this time there can be no assurance that the Company will obtain forgiveness of the loan in whole or in part.

Loan recipients may elect an eight week or 24 week forgiveness period and the repayment period begins on the date which the amount of forgiveness is determined. In the event that a loan recipient has not applied for forgiveness within 10 months of the end of its covered forgiveness period, the loan recipient must begin making principal and interest payments on that date. As of March 31, 2021 and September 30, 2020, the full amount of the PPP Loan is classified as a note payable within the Company's consolidated balance sheets.

5. Leases

The Company adopted ASC 842 - Leases on October 1, 2019. Through September 30, 2019, the Company's leases consisted of leased office space under various operating leases with terms of one year or less. These leases qualified as short-term leases and as such, there was no cumulative impact from the adoption of ASC 842.

In December 2019, the Company entered a 36-month lease for office space in Princeton, New Jersey, which commenced February 1, 2020. At that time, the Company terminated its existing month-to-month leases for office space.

The components of lease expense for the six months ended March 31, 2021 are as follows:

Lease expense		
Operating lease expense		\$ 51,078
Short-term lease expense		 12,996
Total lease cost		\$ 64,074
		 -
	11	

Sonnet BioTherapeutics Holdings, Inc. Notes to Unaudited Interim Consolidated Financial Statements

At March 31, 2021, the weighted-average remaining lease term was 1.83 years and the weighted average discount rate was 12%.

Cash paid for amounts included in the measurement of lease liabilities:

Operating cash flow from operating loss	\$ 50,442
Future minimum lease payments under non-cancellable leases at March 31, 2021 are as follows:	
Fiscal year	
2021 (excluding the six months ended March 31, 2021)	\$ 51,075
2022	103,440
2023	34,695
Total undiscounted lease payments	 189,210
Less: imputed interest	(21,524)
	\$ 167,686
Total lease liabilities	 <u> </u>

6. Commitments and Contingencies

Legal Proceedings

From time to time, the Company is a party to various lawsuits, claims, and other legal proceedings that arise in the ordinary course of its business. While the outcomes of these matters are uncertain, management does not expect that the ultimate costs to resolve these matters will have a material adverse effect on the Company's financial position, results of operations, or cash flows.

The Company has entered into a Discovery Collaboration Agreement (the "Collaboration Agreement") with XOMA (US) LLC ("XOMA"), pursuant to which XOMA granted to Sonnet a non-exclusive, non-transferrable license and/or right to use certain materials, technologies and related information related to discovery, optimization and development of antibodies and related proteins and to develop and commercialize products thereunder. Sonnet is obligated to make contingent milestone payments to XOMA totaling \$3.8 million on a product-by-product basis upon the achievement of certain development and approval milestones related to a product. Sonnet has also agreed to pay XOMA low single-digit royalties on net sales of products sold by Sonnet. Royalties on each product are payable on a country-by-country basis until the later of (i) a specified period of time after the first commercial sale, and (ii) the date of expiration of the last valid claim in the last-to-expire of the issued patents covered by the Collaboration Agreement.

The Company has entered into a License Agreement (the "ARES License Agreement") with Ares Trading, a wholly-owned subsidiary of Merck KGaA ("ARES"). Under the terms of the ARES License Agreement, ARES has granted the Company a sublicensable, exclusive, worldwide, royalty-bearing license on proprietary patents to research, develop, use and commercialize products using atexakin alfa ("Atexakin"), a low dose formulation of human interleukin-6 in peripheral neuropathies and vascular complications. Pursuant to the ARES License Agreement, the Company will pay ARES high single-digit royalties on net sales of products sold by the Company. Royalties are payable on a product-by-product and country-by-country basis until the later of (i) a specified period of time after the first commercial sale in such country, and (ii) the last date on which such product is covered by a valid claim in such country.

12

Sonnet BioTherapeutics Holdings, Inc. Notes to Unaudited Interim Consolidated Financial Statements

Employment Agreements

The Company has entered into employment contracts with its officers and certain employees that provide for severance and continuation of benefits in the event of termination of employment either by the Company without cause or by the employee for good reason, both as defined in the contract. In addition, in the event of termination of employment following a change in control, as defined, either by the Company without cause or by the employee for good reason, any unvested portion of the employee's initial stock option grant becomes immediately vested.

7. Stockholders' Equity

Common stock

During the six months ended March 31, 2021, the Company sold common stock of 4,021,561 shares under the at-the-market sales agreement discussed in Note 1 for gross proceeds of an aggregate of \$10.6 million and net proceeds of \$10.2 million.

During the six months ended March 31, 2020, the Company sold 186,075 shares of common stock for net proceeds of \$4.0 million. In addition, the Company issued 8,526 shares of common stock and warrants to purchase 4,262 upon conversion of outstanding promissory notes with an outstanding principal balance of \$0.2 million at the time of conversion.

Common stock warrants

As of March 31, 2021, the following equity-classified warrants and related terms were outstanding:

	Warrants Outstanding	 Exercise Price	Expiration Date
Warrants	105,812	\$ 29.32	October 1, 2022 - March 10, 2023
Chanticleer warrants	17,760	\$ 58.50 - \$91.00	April 30, 2027 - December 17, 2028
Series B warrants	42,373	\$ 0.0001	April 16, 2025
Series C warrants	11,329,463	\$ 3.19	October 16, 2025
	11,495,408		

During the six months ended March 31, 2021, the Series B warrant holders exercised 23,863 warrants for proceeds of \$2. An additional 2,242,427 of Series B warrants were net share settled, resulting in the issuance of 2,242,339 shares of common stock.

During the six months ended March 31, 2021, the Chanticleer warrants to purchase 186,161 shares of common stock with an exercise price of \$0.01 per share were net share settled, resulting in the issuance of 185,422 shares of common stock.

13

Sonnet BioTherapeutics Holdings, Inc. Notes to Unaudited Interim Consolidated Financial Statements

8. Share-Based Compensation

In April 2020, the Company adopted the 2020 Omnibus Equity Incentive Plan (the "Plan"). The total number of shares authorized under the Plan as of March 31, 2021 was 653,846, all of which have been granted as of March 31, 2021. The Plan increases the amount of shares issuable under the Plan by four percent of the outstanding shares of common stock at each January 1, each year. The Plan permits the granting of share-based awards, including stock options, restricted stock units and awards, stock appreciation rights and other types of awards as deemed appropriate, in each case, in accordance with the terms of the Plan. The terms of the awards are determined by the Company's Board of Directors.

Restricted Stock Units

In July of 2020, 653,846 restricted stock units ("RSUs") were granted, 50% of which vest on April 2, 2021 and the remaining 50% vest on April 2, 2022. Any unvested RSUs will be forfeited upon termination of services. The fair value of an RSU is equal to the fair market value of the Company's common stock on the date of grant. RSU expense is amortized straight-line over the vesting period.

The Company recorded share-based compensation expense associated with the RSUs in its accompanying statements of operations.

Six Months Ended March 31, 2021

Research and development \$ 211,389

General and administrative	528,721
	\$ 740.110

The following table summarizes RSU activity under the Plan:

	RSU	W	eighted Average Grant Date Fair Value
Unvested balance at September 30, 2020	653,845	\$	3.63
Granted	_		_
Unvested balance at March 31, 2021	653,845	\$	3.63

As of March 31, 2021, total unrecognized compensation expense relating to unvested RSUs granted was \$1.3 million, which is expected to be recognized over a period of 1.0 year.

9. Subsequent Event

The Company has evaluated subsequent events and there are no items requiring disclosure except the following:

On May 2, 2021, the Company entered into a License Agreement (the "Agreement") with New Life Therapeutics PTE, LTD ("New Life"). Sonnet granted New Life an exclusive license to develop and commercialize pharmaceutical preparations containing a specific recombinant human interleukin-6, SON-080 (the "Compound") (such preparations, the "Products") for the prevention, treatment or palliation of diabetic peripheral neuropathy in humans (the "DPN Field") in Malaysia, Singapore, Indonesia, Thailand, Philippines, Vietnam, Brunei, Myanmar, Lao PDR and Cambodia (the "Exclusive Territory").

New Life will bear the cost of, and be responsible for conducting clinical studies and additional non-clinical studies and other developmental and regulatory activities for and commercializing Products in the DPN Field.

New Life will pay the Company a \$500,000 upfront cash payment and is obligated to pay a deferred license fee of an additional \$1,000,000 at the time of the satisfaction of certain milestones as well as potential additional milestone payments to the Company up to \$19,000,000 subject to the achievement of certain development and commercialization milestones. In addition, during the Royalty Term, New Life is obligated to pay the Company tiered double digit royalties ranging from 12% to 30% based on annual net sales of Products in the Territory.

14

ITEM 2: MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following discussion and analysis of our financial condition and results of operations should be read together with our financial statements and the related notes and the other financial information included elsewhere in this Quarterly Report. This discussion contains forward-looking statements that involve risks and uncertainties. Our actual results could differ materially from those anticipated in these forward-looking statements as a result of various factors, including those discussed below and elsewhere in this Quarterly Report, particularly those under "Risk Factors."

CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS

This report on Form 10-Q contains forward-looking statements made pursuant to the safe harbor provisions of the Private Securities Litigation Reform Act of 1995 under Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended. Forward-looking statements include statements with respect to our beliefs, plans, objectives, goals, expectations, anticipations, assumptions, estimates, intentions and future performance, and involve known and unknown risks, uncertainties and other factors, which may be beyond our control, and which may cause our actual results, performance or achievements to be materially different from future results, performance or achievements expressed or implied by such forward-looking statements. All statements other than statements of historical fact are statements that could be forward-looking statements. You can identify these forward-looking statements through our use of words such as "may," "can," "anticipate," "assume," "should," "indicate," "would," "believe," "contemplate," "expect," "seek," "estimate," "continue," "plan," "point to," "project," "predict," "could," "intend," "target," "potential" and other similar words and expressions of the future.

There are a number of important factors that could cause the actual results to differ materially from those expressed in any forward-looking statement made by us. These factors include, but are not limited to:

- our lack of operating history and history of operating losses;
- our need for significant additional capital and our ability to satisfy our capital needs;
- our ability to complete required clinical trials of our products and obtain approval from the FDA or other regulatory agents in different jurisdictions;
- the potential impact of the recent COVID-19 pandemic on our operations, including on our clinical development plans and timelines;
- our ability to maintain or protect the validity of our patents and other intellectual property;
- our ability to retain key executive members;
- our ability to internally develop new inventions and intellectual property;
- interpretations of current laws and the passages of future laws;
- acceptance of our business model by investors;
- the accuracy of our estimates regarding expenses and capital requirements; and
- our ability to adequately support growth.

The foregoing does not represent an exhaustive list of matters that may be covered by the forward-looking statements contained herein or risk factors that we are faced with that may cause our actual results to differ from those anticipate in our forward-looking statements. Please see "Risk Factors" for additional risks which could adversely impact our business and financial performance.

All forward-looking statements are expressly qualified in their entirety by this cautionary notice. You are cautioned not to place undue reliance on any forward-looking statements, which speak only as of the date of this report or the date of the document incorporated by reference into this report. We have no obligation, and expressly disclaim any obligation, to update, revise or correct any of the forward-looking statements, whether as a result of new information, future events or otherwise. We have expressed our expectations, beliefs and projections in good faith and we believe they have a reasonable basis. However, we cannot assure you that our expectations, beliefs or projections will result or be achieved or accomplished.

Overview

Sonnet BioTherapeutics Holdings, Inc. ("Sonnet Holdings" or ," "we," "us," "our," or the "Company"), is a clinical stage, oncology-focused biotechnology company with a proprietary platform for innovating biologic medicines of single- or bi-specific action. Known as F_HAB^{TM} (Fully Human Albumin Binding), the technology utilizes a fully human single chain antibody fragment (scFv) that binds to and "hitch-hikes" on human serum albumin (HSA) for transport to target tissues. Our lead proprietary asset, SON-1010, is a fully human version of Interleukin 12 (IL-12), covalently linked to the F_HAB construct, for which we intend to pursue clinical development in solid tumor indications, including non-small cell lung cancer and head and neck cancer. Sonnet has completed a nonhuman primate (NHP) GLP toxicity study with SON-1010 and is preparing an Investigational New Drug (IND) application for submission to the FDA with the goal of initiating a Phase 1 clinical trial during the second half of 2021. The Company acquired the global development rights to our most advanced compound, a fully human version of Interleukin 6 (IL-6), in April 2020. Going forward, we will exclusively refer to this candidate as SON-080, for its target indications of Chemotherapy-Induced Peripheral Neuropathy (CIPN) and Diabetic Peripheral Neuropathy (DPN), the latter of which had previously been known as the SON-081 program. Sonnet intends to file an IND for a US Phase 1b/2a pilot-scale efficacy study with SON-080 in CIPN during the second half of 2021. Pursuant to a license agreement the Company entered with New Life in May 2021, New Life will be responsible for leading the development program for SON-080 in DPN with the objective of initiating an ex-US Phase 1b/2a pilot-scale efficacy study during the second half of 2021.

We have incurred recurring operating losses and negative cash flows since inception. Our ability to generate product or licensing revenue sufficient to achieve profitability will depend heavily on the successful development and eventual commercialization of one or more of our current or future product candidates. Our net losses were \$11.8 million and \$5.0 million for the six months ended March 31, 2021 and 2020, respectively. As of March 31, 2021, we had cash of \$6.7 million. We expect to continue to incur significant expenses and increasing operating losses for at least the next several years. We expect that our expenses and capital requirements will increase substantially in connection with our ongoing activities, particularly if and as we:

- conduct additional clinical trials for product candidates;
- continue to discover and develop additional product candidates;
- acquire or in-license other product candidates and technologies;
- maintain, expand and protect our intellectual property portfolio;
- hire additional clinical, scientific and commercial personnel;
- establish a commercial manufacturing source and secure supply chain capacity sufficient to provide commercial quantities of any product candidates for which we may obtain regulatory approval;
- seek regulatory approval for product candidates that successfully complete clinical trials;
- establish a sales, marketing and distribution infrastructure to commercialize any products for which we may obtain regulatory approval; and

16

• add operational, financial and management information systems and personnel, including personnel to support our product development and planned future commercialization efforts, as well as to support our operation as a public reporting company.

We will not generate revenue from product sales, if any, unless and until we receive licensing revenue and/or successfully complete clinical development and obtain regulatory approval for our product candidates. If we obtain regulatory approval for any of our product candidates and do not enter into a commercialization partnership, we expect to incur significant expenses related to developing our internal commercialization capability to support product sales, marketing and distribution. As a result of the Merger, as described below, we will continue to incur significant costs associated with operating as a public company.

As a result, we will need substantial additional funding to support our continuing operations and pursue our growth strategy. Until such time as we can generate significant revenue from product sales, if ever, we expect to finance our operations through the sale of equity, debt financings or other capital sources, which may include collaborations with other companies or other strategic transactions. We may not be able to raise additional funds or enter into such other agreements or arrangements when needed on favorable terms, or at all. If we fail to raise capital or enter into such agreements as and when needed, we may have to significantly delay, reduce or eliminate the development and commercialization of one or more of our product candidates or delay our pursuit of potential in-licenses or acquisitions.

Because of the numerous risks and uncertainties associated with product development, we are unable to predict the timing or amount of increased expenses or when or if we will be able to achieve or maintain profitability. Even if we are able to generate product sales, we may not become profitable. If we fail to become profitable or are unable to sustain profitability on a continuing basis, then we may be unable to continue our operations at planned levels and be forced to reduce or terminate operations.

Since our inception in 2015, we have devoted substantially all of our efforts and financial resources to organizing and staffing the Company, business planning, raising capital, acquiring or discovering product candidates and securing related intellectual property rights and conducting discovery, research and development activities for product candidates. We do not have any products approved for sale and have not generated any revenue from product sales. We have funded our operations to date primarily with proceeds from sales of common stock, warrants and proceeds from the issuance of convertible debt.

Recent Events

New Life Therapeutics License Agreement

On May 2, 2021, we entered into a License Agreement (the "Agreement") with New Life Therapeutics PTE, LTD., a company organized under the laws of Singapore ("New Life"). Pursuant to the Agreement, we granted New Life an exclusive license (with the right to sublicense) to develop and commercialize pharmaceutical preparations containing a specific recombinant human interleukin-6, SON-080 (or any derivatives, fragments or conjugates thereof) (the "Compound") (such preparations, the "Products") for the prevention, treatment or palliation of diabetic peripheral neuropathy in humans (the "DPN Field") in Malaysia, Singapore, Indonesia, Thailand, Philippines, Vietnam, Brunei, Myanmar, Lao PDR and Cambodia (the "Exclusive Territory"). New Life may exercise the option to expand (1) the field of the exclusive license to include the prevention, treatment or palliation of chemotherapy-induced peripheral neuropathy in humans (the "CIPN Field"), which option is non-exclusive and will expire on December 31, 2021; and/or (2) the territorial scope of the license to include the People's Republic of China, Hong Kong and/or India, which option is exclusive and will also expire on December 31, 2021. We are excluded from developing, using, selling or otherwise commercializing any Compounds or Products for use in the DPN Field in the Exclusive Territory during the term of the Agreement.

We retain all rights to manufacture Compounds and Products anywhere in the world. We and New Life shall enter into a follow-on supply agreement pursuant to which we shall supply to New Life Products for development and commercialization thereof in the DPN Field (and the CIPN Field, if applicable) in the Exclusive Territory on terms to be negotiated by the parties.

Pursuant to the terms of the Agreement, New Life will bear the cost of, and be responsible for, among other things, conducting clinical studies and additional non-clinical studies (if any, subject to both parties' approval), preparing and filing applications for regulatory approval and undertaking other developmental and regulatory activities for and commercializing Products in the DPN Field (and the CIPN Field, if applicable) in the Exclusive Territory. New Life will own and maintain all regulatory filings and approvals for Products in the Exclusive Territory.

In consideration of the license and other rights granted by us, New Life will pay us, within 30 days of the date of the Agreement, a \$500,000 upfront cash payment and is obligated to pay a deferred license fee of an additional \$1,000,000 at the time of the satisfaction of certain milestones as well as potential additional milestone payments to us totaling up to \$19,000,000 subject to the achievement of certain development and commercialization milestones. In addition, during the Royalty Term (as defined below), New Life is obligated to pay us tiered double digit royalties ranging from 12% to 30% based on annual net sales of Products in the Territory. The "Royalty Term" means, on a Product-by-Product and a country-by-country basis in the Exclusive Territory, the period commencing on the date of the first commercial sale (subject to certain conditions) of such Product in such country in the Exclusive Territory and continuing until New Life ceases commercialization of such Product in the DIPN Field (or CIPN Field, if applicable). In the event New Life (i) files for an initial public offering or (ii) is subject to a Change of Control, the royalty obligations may be converted to equity subject to mutual agreement of the parties.

In addition, New Life shall pay to us a percentage, in the double digits, of all revenue received through sub-licensing of each Product, subject to certain exclusions.

Merger

On April 1, 2020, Chanticleer Holdings, Inc ("Chanticleer"), now known as Sonnet BioTherapeutics Holdings, Inc, completed its merger transaction (the "Merger") with Sonnet BioTherapeutics, Inc. ("Sonnet"), in accordance with the terms of the Agreement and Plan of Merger, dated as of October 10, 2019, as amended on February 7, 2020 (the "Merger Agreement"). Chanticleer shares of common stock traded on the Nasdaq Capital Market through close of Business on Tuesday, March 31, 2020 under the ticker symbol "BURG". We commenced trading on the Nasdaq Capital Market, under the ticker symbol "SONN" on April 2, 2020.

Immediately following the Merger, Sonnet became a wholly-owned subsidiary of Sonnet Holdings. For accounting purposes, Sonnet is considered to be the acquiring company and the Merger has been accounted for as a reverse acquisition and recapitalization with Sonnet being treated as the accounting acquirer. As such, the financial information prior to the Merger relate solely to Sonnet. Subsequent to the Merger, the consolidated financial statements relate to the consolidated entities of the Company.

Relief Acquisition

In August 2019, Sonnet executed a Share Exchange Agreement with Relief Therapeutics Holdings SA ("Relief Holdings"), in which Sonnet agreed to acquire the outstanding shares of Relief Therapeutics SA ("Relief"), a wholly-owned subsidiary of Relief Holdings, by issuing common stock of Sonnet. Sonnet assumed the development of Relief's asset, atexakin alfa, together with its proprietary experimental drugs. The acquisition of Relief closed on April 1, 2020 and Relief is now a wholly-owned subsidiary of Sonnet.

18

COVID-19 Pandemic

In December 2019, a novel strain of coronavirus, COVID-19, was reported to have surfaced in Wuhan, China and on March 11, 2020 was declared a pandemic by the World Health Organization. To date, many countries around the world have imposed quarantines and restrictions on travel and mass gatherings to slow the spread of COVID-19 and have closed non-essential businesses. As countries and state and local jurisdictions continue to put restrictions in place, our ability to continue to operate our business may also be limited. Such events may result in a period of business, supply and drug product manufacturing disruption, and in reduced operations, any of which could materially affect our business, financial condition and results of operations.

This pandemic or outbreak could result in difficulty securing clinical trial site locations, CROs, and/or trial monitors and other critical vendors and consultants supporting the trial. In addition, outbreaks or the perception of an outbreak near a clinical trial site location could impact our ability to enroll patients. These situations, or others associated with COVID-19, could cause delays in our clinical trial plans and could increase expected costs, all of which could have a material adverse effect on our business and its financial condition.

In particular, although our CIPN program with SON-080 continues to progress forward, the COVID-19 pandemic has impacted workflow at our contract research partners such that we now estimate delays pushing a trial initiation into 2021 from our previous plan of late 2020.

While the potential economic impact brought by, and the duration of, COVID-19 may be difficult to assess or predict, a widespread pandemic could result in significant disruption of global financial markets, reducing our ability to access capital, which could in the future negatively affect our liquidity. In addition, a recession or market correction resulting from the spread of COVID-19 could materially affect our business and the value of our common shares.

The COVID-19 outbreak may also affect the ability of our staff and the parties we work with to carry out our non-clinical, clinical, and drug manufacturing activities. We rely or may in the future rely on clinical sites, investigators and other study staff, consultants, independent contractors, contract research organizations and other third-party service providers to assist us in managing, monitoring and otherwise carrying out our nonclinical studies and clinical trials. We also rely or may in the future rely on consultants, independent contractors, contract manufacturing organizations, and other third-party service providers to assist us in managing, monitoring and otherwise carrying out our API production, formulation, and drug manufacturing activities. COVID-19 may affect the ability of any of these external people, organizations, or companies to devote sufficient time and resources to our programs or to travel to perform work for us.

Potential negative impacts of the COVID-19 outbreak on the conduct of current or future clinical studies include delays in gaining feedback from regulatory agencies, starting new clinical studies, and recruiting subjects to studies that are enrolling. The potential negative impacts also include inability to have study visits at study sites, incomplete collection of safety and efficacy data, and higher rates of drop-out of subjects from ongoing studies, delays in site entry of study data into the data base, delays in monitoring of study data because of restricted physical access to study sites, delays in site responses to queries, delays in data-base lock, delays in data analyses, delays in time to top-line data, and delays in completing study reports. New or worsening COVID-19 disruptions or restrictions could have the potential to further negatively impact our non-clinical studies, clinical trials, and drug manufacturing activities.

Operating Expenses

Research and Development Expenses

Research and development expenses consist primarily of costs incurred in connection with the discovery and development of our product candidates. We expense research and development costs as incurred and such costs include:

- employee-related expenses, including salaries, share-based compensation and related benefits, for employees engaged in research and development functions;
- expenses incurred in connection with the preclinical and clinical development of our product candidates, including under agreements with third parties, such as
 consultants and clinical research organizations;
- the cost of manufacturing drug products for use in our preclinical studies and clinical trials, including under agreements with third parties, such as consultants and contract manufacturing organizations;
- · facilities, depreciation and other expenses, which include direct or allocated expenses for rent and maintenance of facilities and insurance;
- costs related to compliance with regulatory requirements; and
- · payments made under third-party licensing agreements.

We recognize external development costs based on an evaluation of the progress to completion of specific tasks using information provided by our service providers. This process involves reviewing open contracts and purchase orders, communicating with their personnel to identify services that have been performed on our behalf, and estimating the level of service performed and the associated cost incurred for the service when we have not yet been invoiced or otherwise notified of actual costs. Nonrefundable advance payments for goods or services to be received in the future for use in research and development activities are recorded as prepaid expenses. Such amounts are recognized as an expense when the goods have been delivered or the services have been performed.

Our direct research and development expenses consist primarily of external costs, such as fees paid to outside consultants, CROs, CMOs and research laboratories in connection with preclinical development, process development, manufacturing and clinical development activities. Our direct research and development expenses also include fees incurred under third-party license agreements. We do not allocate employee costs and costs associated with discovery efforts, laboratory supplies and facilities, including depreciation or other indirect costs, to specific product candidates because these costs are deployed across multiple programs and as such, are not separately classified. We use internal resources primarily to conduct its research and discovery as well as for managing preclinical development, process development, manufacturing and clinical development activities. These employees work across multiple programs and therefore, we do not track its costs by product candidate.

20

We expect our research and development expense will increase for the foreseeable future as we attempt to advance development of our product candidates. The successful development of our product candidates is highly uncertain. At this time, we cannot reasonably estimate or know the nature, timing and costs of the efforts that will be necessary to complete the remainder of the development of our current pipeline or any future product candidates we may develop due to the numerous risks and uncertainties associated with clinical development, including risk and uncertainties related to:

- the timing and progress of preclinical and clinical development activities;
- the number and scope of preclinical and clinical programs that we decide to pursue;
- Our ability to maintain our current research and development programs and to establish new ones;
- establishing an appropriate safety profile with investigational new drug-enabling studies;
- successful patient enrollment in, and the initiation and completion of, clinical trials;
- the successful completion of clinical trials with safety, tolerability and efficacy profiles that are satisfactory to the FDA or any comparable foreign regulatory authority;
- the receipt of regulatory approvals from applicable regulatory authorities;
- the timing, receipt and terms of any marketing approvals from applicable regulatory authorities;
- our ability to establish new licensing or collaboration arrangements;
- establishing agreements with third-party manufacturers for clinical supply for our clinical trials and commercial manufacturing, if any of our product candidates is approved;
- development and timely delivery of clinical-grade and commercial-grade drug formulations that can be used in our clinical trials and for commercial launch;
- obtaining, maintaining, defending and enforcing patent claims and other intellectual property rights;
- launching commercial sales of product candidates, if approved, whether alone or in collaboration with others;
- maintaining a continued acceptable safety profile of the product candidates following approval; and
- the potential impact of COVID-19 on operations which may affect among other things, the timing of clinical trials, availability of raw materials, and the ability to access and secure testing facilities.

21

A change in the outcome of any of these variables with respect to the development of our product candidates could significantly change the costs and timing associated with the development of that product candidate. We may never succeed in obtaining regulatory approval for any of our product candidates.

General and Administrative Expenses

General and administrative expenses consist primarily of salaries and related costs for personnel, including share-based compensation, in executive, finance and administrative

functions. General and administrative expenses also include direct and allocated facility-related costs as well as professional fees for legal, patent, consulting, accounting, and audit services.

Our general and administrative expenses will increase in the future as we increase our headcount to support continued research activities and development of product candidates. We will continue to incur increased accounting, audit, legal, regulatory, compliance and director and officer insurance costs as well as investor and public relations expenses associated with being a public company.

Foreign exchange loss

Foreign exchange loss consists of exchange rate changes on transactions denominated in currencies other than the U.S. dollar.

Results of Operations

Comparison of the three months ended March 31, 2021 and 2020

The following table summarizes our results of operations for the three months ended March 31, 2021 and 2020:

Three Months Ended March 31, 2021 2020 Change Operating expenses Research and development \$ 3,840,399 1,302,515 2,537,884 General and administration 988.258 2,196,632 1,208,374 Loss from operations (6.037.031)(2,510,889)(3.526.142)Foreign exchange gain (loss) (2.076)14 142 (16.218)Net loss (6,039,107)(3.542.360)(2,496,747)

Research and Development Expenses

Research and development expenses were \$3.8 million for the three months ended March 31, 2021, compared to \$1.3 million for the three months ended March 31, 2020. The increase of \$2.5 million was primarily due to increased expenditures for the development of the cell line for IL12-FHAB and IL12-FHAB-IL15 and increased costs for research and development activities due to the acquisition of Relief and an increase in payroll and share-based compensation expense as we expanded our operations.

22

General and Administrative Expenses

General and administrative expenses were \$2.2 million for the three months ended March 31, 2021, compared to \$1.2 million for the three months ended March 31, 2020. The increase of \$1.0 million was primarily due to an increase in insurance expense of approximately \$0.2 million related to directors and officer's insurance, and an increase in payroll and share-based compensation expense of \$0.4 million to support our expanded operations.

Comparison of the six months ended March 31, 2021 and 2020

The following table summarizes our results of operations for the six months ended March 31, 2021 and 2020:

	Six Months Ended March 31,					
		2021		2020		Change
Operating expenses						
Research and development	\$	7,706,407	\$	2,710,663	\$	4,995,744
General and administration		4,194,617		2,269,280		1,925,337
Loss from operations		(11,901,024)		(4,979,943)		(6,921,081)
Foreign exchange gain (loss)		(15,323)		14,142		(29,465)
Net loss	\$	(11,916,347)	\$	(4,965,801)	\$	(6,950,546)

Research and Development Expenses

Research and development expenses were \$7.7 million for the six months ended March 31, 2021, compared to \$2.7 million for the six months ended March 31, 2020. The increase of \$5.0 million was primarily due to increased expenditures for the development of the cell line for IL12-FHAB and IL12-FHAB-IL15 and increased costs for research and development activities due to the acquisition of Relief and an increase in payroll and share-based compensation expense as we expanded our operations.

General and Administrative Expenses

General and administrative expenses were \$4.2 million for the six months ended March 31, 2021, compared to \$2.3 million for the six months ended March 31, 2020. The increase of \$1.9 million was primarily due to an increase in insurance expenses related to directors and officer's insurance, and an increase in payroll and share-based compensation expense as we expanded our operations to support our overall business objectives.

Liquidity and Capital Resources

Since inception, we have not generated any significant revenue from any sources, including from product sales, and have incurred recurring losses and negative cash flows from operations. We have funded operations to date primarily with proceeds from sales of common stock, warrants and proceeds from the issuance of convertible debt. Although we entered into the Agreement with New Life, all of the potential proceeds from the Agreement, except for the upfront payment that is due within 30 days of the execution of the Agreement, are contingent on various milestones or other criteria being achieved (see the discussion of the Agreement above in the section titled "Recent Events—New Life Therapeutics License Agreement").

Six Months Ended March 31.

	202	2021 2020		2020
Net cash used in operating activities	\$ (1)	0,769,285)	\$	(3,551,102)
Net cash used in investing activities		_		(65,265)
Net cash provided by financing activities	1	0,158,193		3,853,569
Net (decrease) increase in cash	\$	(611,092)	\$	237,202

Operating Activities

During the six months ended March 31, 2021, we used \$10.8 million of cash in operating activities which was primarily attributable to our net loss of \$11.9 million. This amount was offset by a \$0.7 million in share-based compensation expense and a net increase of \$0.7 million in accounts payable and accrued expenses primarily attributable increased research and development efforts.

During the six months ended March 31, 2020, we used \$3.6 million of cash in operating activities. Cash used in operating activities reflected our net loss of \$5.0 million offset by a \$1.5 million net increase in accounts payable and accrued expenses primarily attributable increased research and development efforts.

Investing Activities

During the six months ended March 31, 2020, we purchased \$22 thousand of office furniture and computer equipment. No purchased of equipment were made during the six months ended March 31, 2021.

Financing Activities

During the six months ended March 31, 2021, we received net proceeds of \$10.2 million from the sale of common stock under our at-the market facility.

During the six months ended March 31, 2020, net cash provided by financing activities was \$3.9 million, primarily consisting of net proceeds of \$4.1 million from the sale of common stock, partially offset by \$0.2 million in net repayments of related-party notes.

Funding Requirements

We expect our expenses to increase substantially in connection with our ongoing activities, particularly as we advance preclinical activities and clinical trials of product candidates in development. In addition, we expect to incur additional costs associated with operating as a public company. The timing and amount of our operating expenditures will depend largely on:

24

- the scope, number, initiation, progress, timing, costs, design, duration, any potential delays, and results of clinical trials and nonclinical studies for our current or future product candidates;
- the clinical development plans we establish for these product candidates;
- the number and characteristics of product candidates and programs that we develop or may in-license;
- the outcome, timing and cost of regulatory reviews, approvals or other actions to meet regulatory requirements established by the FDA and comparable foreign regulatory authorities, including the potential for the FDA or comparable foreign regulatory authorities to require that we perform more studies for our product candidates than those that we currently expect;
- our ability to obtain marketing approval for product candidates;
- the cost of filing, prosecuting, defending and enforcing patent claims and other intellectual property rights covering our product candidates;
- our ability to maintain, expand and defend the scope of our intellectual property portfolio, including the cost of defending intellectual property disputes, including patent infringement actions brought by third parties against us or our product candidates;
- $\bullet \quad \text{the cost and timing of completion of commercial-scale outsourced manufacturing activities with respect to product candidates;} \\$
- our ability to establish and maintain licensing, collaboration or similar arrangements on favorable terms and whether and to what extent we retain development or commercialization responsibilities under any new licensing, collaboration or similar arrangement;
- the cost of establishing sales, marketing and distribution capabilities for any product candidates for which we may receive regulatory approval in regions where we choose to commercialize our products on our own:
- the success of any other business, product or technology that the we acquire or in which we invest;
- the costs of acquiring, licensing or investing in businesses, product candidates and technologies;
- our need and ability to hire additional management and scientific and medical personnel;
- the costs to operate as a public company in the United States, including the need to implement additional financial and reporting systems and other internal systems and infrastructure for our business;
- market acceptance of our product candidates, to the extent any are approved for commercial sale;
- the effect of competing technological and market developments; and
- the potential impact of the COVID-19 pandemic on our clinical trials and operations.

Until such time, if ever, as the we can generate substantial product revenue, we expect to finance our cash needs through a combination of equity offerings, debt financings, collaborations, strategic alliances, and marketing, distribution or licensing arrangements with third parties. To the extent that we raise additional capital through the sale of equity or convertible debt securities, the ownership interest of ours may be materially diluted, and the terms of such securities could include liquidation or other preferences that adversely affect the rights of our stockholders. Debt financing and preferred equity financing, if available, may involve agreements that include restrictive covenants that limit our ability to take specified actions, such as incurring additional debt, making capital expenditures or declaring dividends. If we raise funds through collaborations, strategic alliances or marketing, distribution or licensing arrangements with third parties, we may have to relinquish valuable rights to technologies, future revenue streams, research programs or product candidates or grant licenses on terms that may not be favorable to us. If we are unable to raise additional funds through equity or debt financings or other arrangements when needed, we may be required to delay, reduce or eliminate product development or future commercialization efforts, sell off assets, or grant rights to develop and market product candidates that we would otherwise prefer to develop and market.

We believe our cash of \$6.7 million at March 31, 2021 will fund our projected operations into July 2021; provided that if we sold the remaining \$5.3 million in gross proceeds available under the Sales Agreement (as defined below), we believes our cash at March 31, 2021 together with the net proceeds therefrom would fund our projected operations into September 2021. We will need to raise significant additional capital in the near term to fund our continuing operations.

We entered into an At-the-Market Sales Agreement with BTIG, LLC ("BTIG") on February 5, 2021 (the "Sales Agreement"). Pursuant to the Sales Agreement, we may offer and sell, from time to time, through BTIG, as sales agent and/or principal, shares of its common stock, having an aggregate offering price of up to \$15,875,000, subject to certain limitations on the amount of common stock that may be offered and sold by us set forth in the Sales Agreement. We are not obligated to make any sales of shares under the Sales Agreement and any determination by us to do so will be dependent, among other things, on market conditions and our capital raising needs. Through March 31, 2021, we sold an aggregate of 4,021,561 shares under the Sales Agreement for gross proceeds of \$10.6 million and net proceeds of \$10.2 million.

Contractual Obligations and Commitments

The following table summarizes our contractual obligations as of December 31, 2020 and the effects that such obligations are expected to have on our liquidity and cash flows in future periods:

	l	Less than 1					Moı	re than 5			
		Year		1 to 3 Years		4 to 5 Years		Years		Total	
Operating Lease (1)	\$	51,075	\$	138,135	\$		\$		\$	189,210	
Debt Obligations (2)	\$	748	\$	125,501					\$	126,249	
Total	\$	51,823	\$	263,636	\$	_	\$	_	\$	315,459	

- (1) Reflects obligations pursuant to our office lease in Princeton, New Jersey.
- (2) Reflects unsecured notes payable issued to certain related parties and a loan under the Payroll Protection Program.

26

In addition to the contracts with payment commitments that we have reflected in the table above, we have entered into other contracts in the normal course of business with certain CROs, CMOs and other third-parties for preclinical research studies and testing, clinical trials and manufacturing services. These contracts do not contain any minimum purchase commitments and are cancelable upon prior notice and as a result, are not included in the table of contractual obligations and commitments above. Payments due upon cancellation consist only of payments for services provided and expenses incurred, including non-cancelable obligations to our service providers, up to the date of cancellation.

Critical Accounting Policies

Our management's discussion and analysis of financial condition and results of operations are based on our consolidated financial statements, which have been prepared in accordance with U.S. GAAP. The preparation of these consolidated financial statements requires us to make estimates and judgments that affect the reported amounts of assets, liabilities, and expenses and the disclosure of contingent assets and liabilities in our financial statements. On an ongoing basis, we evaluate our estimates and judgments, including those related to the accrual for research and development expenses. We base our estimates on historical experience, known trends and events, and various other factors that are believed to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. Actual results may differ from these estimates under different assumptions or conditions.

While our significant accounting policies are described in more detail in the notes to the interim consolidated financial statements included elsewhere in this Form 10-Q, we believe that the following accounting policies are those most critical to the judgments and estimates used in the preparation of the financial statements.

Research and development expenses

Research and development expense consist primarily of costs incurred in connection with the development of the our product candidates. We expense research and development costs as incurred.

At the end of each reporting period, we compare payments made to third-party service providers to the estimated progress toward completion of the applicable research or development objectives. Such estimates are subject to change as additional information becomes available. Depending on the timing of payments to the service providers and the progress that we estimate has been made as a result of the service provided. We may record net prepaid or accrued expense relating to these costs. As of March 31, 2021, we did not make any material adjustments to our prior estimates of accrued research and development expenses.

Off-Balance Sheet Arrangements

We do not have any relationships with unconsolidated entities or financial partnerships, including entities sometimes referred to as structured finance or special purpose entities that were established for the purpose of facilitating off-balance sheet arrangements or other contractually narrow or limited purposes. We do not engage in off-balance sheet financing arrangements. In addition, we do not engage in trading activities involving non-exchange traded contracts. We therefore believe that we are not materially exposed to any financing, liquidity, market or credit risk that could arise if it had engaged in these relationships.

Recently Issued Accounting Pronouncements

A description of recently issued accounting pronouncements that may potentially impact the our financial position and results of operations is disclosed in Note 2 to the interim consolidated financial statements included elsewhere in this Form 10-Q.

27

ITEM 4: CONTROLS AND PROCEDURES

Evaluation of Disclosure Controls and Procedures

We evaluated, under the supervision and with the participation of the principal executive officer and principal financial officer, the effectiveness of the design and operation of our disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) under the Securities and Exchange Act of 1934, as amended ("Exchange Act")) as of March 31, 2021, the end of the period covered by this report on Form 10-Q. Based on this evaluation, our Chairman, President and Chief Executive Officer (principal executive officer) and our Chief Financial Officer (principal financial officer) have concluded that our disclosure controls and procedures were effective at the reasonable assurance level at March 31, 2021.

Disclosure controls and procedures are designed to provide reasonable assurance that information required to be disclosed by us in the reports that we file or submit under the Exchange Act (i) is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms, and (ii) is accumulated and communicated to management, including the principal executive officer and principal financial officer, as appropriate, to allow timely decisions regarding required disclosures. In designing and evaluating the disclosure controls and procedures, management recognizes that any controls and procedures, no matter how well designed and operated, can provide only reasonable assurance of achieving the desired control objectives. Due to the inherent limitations of control systems, not all misstatements may be detected. These inherent limitations include the realities that judgments in decision-making can be faulty and that breakdowns can occur because of a simple error or mistake. Additionally, controls can be circumvented by the individual acts of some persons, by collusion of two or more people, or by management override of the control. Controls and procedures can only provide reasonable, not absolute, assurance that the above objectives have been met.

Changes in Internal Control over Financial Reporting

There were no changes in our internal control over financial reporting during the three months ended March 31, 2021 that have materially affected, or are reasonably likely to materially affect, the Company's internal control over financial reporting.

28

PART II - OTHER INFORMATION

ITEM 1: LEGAL PROCEEDINGS

We are subject to various legal proceedings from time to time in the ordinary course of business, which may not be required to be disclosed under this Item 1. For the three-month period ending March 31, 2021 covered by this Quarterly Report, there have been no reportable legal proceedings or material developments to previously reported legal proceedings.

ITEM 1A: RISK FACTORS

As a smaller reporting company, we are not required to provide the information required by this item. However, we direct you to the risk factors included in the Risk Factors section in our Annual Report on Form 10-K for the year ended September 30, 2020 filed with the Securities and Exchange Commission on December 17, 2020 and the updates set forth below.

The Agreement with New Life is important to our business. If we or New Life fail to adequately perform under the Agreement, or if we or New Life terminate the Agreement, the development and commercialization of SON-080 for the treatment of DPN in the Exclusive Territory would be delayed or terminated and our business would be adversely affected.

On May 2, 2021, we entered into the Agreement with New Life, pursuant to which we granted to New Life an exclusive license to develop and commercialize pharmaceutical preparations containing a specific recombinant human interleukin-6, SON-080 (or any derivatives, fragments or conjugates thereof) (the "Compound") (such preparations, the "Products") for the prevention, treatment or palliation of diabetic peripheral neuropathy ("DPN") in humans in the Exclusive Territory. Our ability to generate revenue under the Agreement will depend in large part on New Life's success in clinical development of SON-080 and success in achieving regulatory approval for, and commercializing SON-080, in the Exclusive Territory. Such efforts are subject to significant uncertainty. We have no control over the resources, time and effort that New Life may devote to the commercialization of SON-080. Any of several events or factors could have a material adverse effect on our ability to generate revenue from New Life's commercialization of SON-080 in the Exclusive Territory. For example, New Life:

- may not complete successfully or the clinical trials necessary for approval of SON-080 for the treatment of DPN in the Exclusive Terriroty;
- may not achieve satisfactory levels of market acceptance and reimbursement by physicians, patients and third-party payers for SON-080 for the treatment of DPN;
- may not compete successfully against other products and therapies for DPN;
- may have to comply with additional requests and recommendations from foreign regulatory authorities;
- may not make all regulatory filings and obtain all necessary approvals from foreign regulatory agencies and all commercially necessary reimbursement approvals;
- may not commit sufficient resources to the marketing and distribution of SON-080, whether for competitive or strategic reasons or otherwise due to a change in business priorities; and
 - may cease to perform its obligations under the terms of the Agreement.

In addition, pursuant to the Agreement, we and New Life have agreed to negotiate in good faith to enter into a supply agreement. There can be no assurance that we will be able to reach mutually agreeable terms on such agreement with New Life, and the absence of agreement on such terms would prevent us from gaining the expected benefit of the Agreement.

The price of our common stock has been and could remain volatile, including recently, and the market price of our common stock may decrease.

The market price of our common stock has historically experienced and may continue to experience significant volatility. From April 2020 through May 14, 2021, the market price of the Company's common stock has fluctuated from a high of \$16.20 per share in April 2020, to a low of \$1.44 per share in May 2021. Market prices for securities of life sciences companies have historically been particularly volatile. The factors that may cause the market price of our common stock to fluctuate include, but are not limited to:

20

- our ability to complete required clinical trials of our products and obtain approval from the FDA or other regulatory agents in different jurisdictions;
- progress, or lack of progress, in developing and commercializing our products;
- favorable or unfavorable decisions about our products from government regulators, insurance companies or other third-party payors;
- our ability to recruit and retain qualified regulatory and research and development personnel;
- changes in investors' and securities analysts' perception of the business risks and conditions of our business;
- changes in our relationship with key collaborators;
- changes in the market valuation or earnings of our competitors or companies viewed as similar to us;

- changes in key personnel;
- depth of the trading market in our common stock;
- changes in our capital structure, such as future issuances of securities or the incurrence of additional debt;
- the granting or exercise of employee stock options or other equity awards;
- realization of any of the risks described under this section titled "Risk Factors"; and
- general market and economic conditions.

In addition, the equity markets have experienced significant price and volume fluctuations that have affected the market prices for the securities of newly public companies for a number of reasons, including reasons that may be unrelated to our business or operating performance. These broad market fluctuations may result in a material decline in the market price of our common stock and you may not be able to sell your shares at prices you deem acceptable. In the past, following periods of volatility in the equity markets, securities class action lawsuits have been instituted against public companies. Such litigation, if instituted against us, could result in substantial cost and the diversion of management attention.

ITEM 2: UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

None.

ITEM 3: DEFAULTS UPON SENIOR SECURITIES

None noted.

ITEM 4: MINE SAFETY DISCLOSURES

Not applicable.

30

ITEM 5: OTHER INFORMATION

Description

None.

Exhibit No

ITEM 6: EXHIBITS

INO.	Description
10.1#	License Agreement, dated May 2, 2021, between Sonnet BioTherapeutics, Inc. and New Life Therapeutics PTE, LTD.
31.1	Certification of the Principal Executive Officer pursuant to Rule 13a-14(a) or Rule 15d-14(a).
31.2	Certification of the Chief Financial Officer pursuant to Rule 13a-14(a) or Rule 15d-14(a).
32.1**	Certification of Principal Executive Officer pursuant to Rule 13a-14(b) or Rule 15d-14(b).
32.2**	Certification of Chief Financial Officer pursuant to Rule 13a-14(b) or Rule 15d-14(b).
101.INS*	XBRL Instance Document
101.SCH*	XBRL Taxonomy Extension Schema Document
101.CAL*	XBRL Taxonomy Extension Calculation Linkbase Document
101.DEF*	XBRL Taxonomy Extension Definition Linkbase Document
101.LAB*	XBRL Taxonomy Extension Label Linkbase Document
101.PRE*	XBRL Taxonomy Extension Presentation Linkbase Document

Filed herewith; portions of the exhibit have been omitted pursuant to Item 601(b)(10) of Regulation S-K. A copy of any omitted portions will be furnished to the Securities and Exchange Commission upon request.

May 17, 2021

Date:

3

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the Registrant has duly caused this Report to be signed on its behalf by the undersigned, thereunto duly authorized on May 17, 2021.

SONNET BIOTHERAPEUTICS HOLDINGS, INC.

By: /s/ Pankaj Mohan

Pankaj Mohan

President and Chief Executive Officer (Principal Executive Officer)

^{*} XBRL (Extensible Business Reporting Language) information is furnished and not filed or a part of a registration statement or prospectus for purposes of Sections 11 or 12 of the Securities Act of 1933, as amended, is deemed not filed for purposes of Section 18 of the Securities Exchange Act of 1934, as amended, and otherwise is not subject to liability under these sections.

^{**} Furnished, not filed.

/s/ Jay Cross
Jay Cross
Chief Financial Officer
(Principal Financial Officer and Principal Accounting Officer)

CERTAIN IDENTIFIED INFORMATION HAS BEEN EXCLUDED FROM THIS EXHIBIT BECAUSE IT IS NOT MATERIAL AND WOULD LIKELY CAUSE COMPETITIVE HARM TO THE REGISTRANT IF PUBLICLY DISCLOSED. THE OMISSIONS HAVE BEEN INDICATED BY "[***]".

LICENSE AGREEMENT

between

SONNET BIOTHERAPEUTICS, INC

and

NEW LIFE THERAPEUTICS PTE, LTD.

Dated as of May 2, 2021

TABLE OF CONTENTS

	PAGES
ARTICLE 1 - DEFINITIONS	5
ARTICLE 2 - GRANT OF LICENSE and option	12
2.1 Grant of License	12
2.2 Grant of Sublicense by Licensee	13
2.3 Non-Compete	13
2.4 No Implied Licenses	13
2.5 Transfer	13
2.6 Territory Expansion Option	13
2.7 Non-Exclusive Option to CIPN Field	13
2.8 Buy Back Option	14
ARTICLE 3 - DEVELOPMENT AND COMMERCIALIZATION	14
3.1 Development of the Product by Licensee	14
3.2 Commercialization	15
3.3 Manufacturing and Supply	15
3.4 Regulatory Filings	15 15
3.5 Diligence 3.6 Annual Reporting	15
3.7 Trademarks	15
5.7 Heddinard	10
ARTICLE 4 - DEVELOPMENT MANAGEMENT	16
4.1 Joint Development Committee	16
4.2 Appointment of Subcommittees, Project Teams and Collaboration Managers	17
4.3 Reports and Minutes	17
4.4 Decision-Making and Dispute Resolution	17
4.5 Dissolution of JDC	18
ARTICLE 5 - Financials	19
The Total of the T	19
5.1 Upfront Payment	19
5.2 Deferred License Fee	19
5.3 Milestones	19
5.4 Royalty; Monetary Third Party Obligations	19
ARTICLE 6 - INVENTIONS, PATENTS and clinical data	21
6.1 Certification Under Drug Price Competition and Patent Restoration Act	21
6.2 Listing of Patents	21
-2-	
6.3 Title to Inventions	21
6.4 Patent Prosecution and Maintenance	21
6.5 Enforcement of Patents	23
6.6 Third Party Actions Claiming Infringement	24
ARTICLE 7 - CONFIDENTIALITY	25
7.1 Confidentiality Obligations	25
7.1 Confidencially Congations 7.2 Publications	26
7.3 Press Releases and Disclosure	26
	20
ARTICLE 8 - REPRESENTATIONS AND WARRANTIES	27

8.1 Repre	esentations and Warranties	27	
8.2 No Fr	urther Representations and Warranties	27	
ARTICLE 9 - IND	EMNIFICATION AND INSURANCE	28	
	mnification by Licensee	28	
9.2 Inden	mnification by SONNET	28	
9.3 No C	Consequential Damages	28	
	fication of Claims; Conditions to Indemnification Obligations	29	
9.5 Insura	ance	29	
ADTICLE 10 TEL	RM AND TERMINATION	29	
ARTICLE 10 - 1E	RM AND TERMINATION	29	
10.1 Term	n of Agreement	29	
	nination of the Agreement for Failure to Use Commercially Reasonable Efforts	29	
	nination for Breach	29	
	nination for Bankruptcy	29	
	ets of Termination	30	
ARTICLE 11 - DIS	SPUTE RESOLUTION - JURISDICTION	31	
11.1 Dispu		31	
	lation to Executive Officers	31	
11.3 Dispute Resolution			
11.4 Injun	active Relief	32	
ARTICLE 12 - MIS	SCELLANEOUS	32	
10.15.1			
	tionship of the Parties	32	
12.2 Assig		32	
	ner Actions	33	
	ts Upon Bankruptcy ounting Procedures	33 33	
12.5 Acco		33	
12.010100	t Majeure	34	
	-3-		
	~		
12.7	No Trademark Rights	34	
12.8	No Sale for Resale	34	
12.9	Conflicting Rights	34	
12.10	Entire Agreement of the Parties; Amendments	34	
12.11	Captions	34	
12.12	Governing Law Notices and Deliveries	35	
12.13 12.14	Notices and Deliveries Waiver	35 35	
12.14	Severability	35	
12.15	Counterparts	35	
12.10	Councipato	33	

LICENSE AGREEMENT

This License Agreement (hereinafter referred to as the "Agreement") is made and effective as of the date of the last signature (the "Effective Date"), by and between SONNET BIOTHERAPEUTICS, INC. (hereinafter referred to as "SONNET"), a company organized under the laws of the state of New Jersey, USA and having its registered office at 100 Overlook Center, Suite 102, Princeton, New Jersey 08540, and NEW LIFE THERAPEUTICS PTE, LTD. (hereinafter referred to as "Licensee"), a company organized under the laws of Singapore and having its registered office at 239 Arcadia Road, #03-03, The Arcadia, Singapore 289845. SONNET and Licensee are each referred to herein as a "Party" and collectively as the "Parties".

WITNESSETH:

WHEREAS, SONNET is engaged, among other activities, in the development of a low dose IL6 Therapeutic for the treatment of DPN (as hereinafter defined) and/or CIPN (as hereinafter defined) the rights to which were acquired by SONNET from Relief Therapeutics, SA pursuant to an assignment of a license agreement dated August 28, 2015 between Relief Therapeutics, SA and ARES TRADING SA, a subsidiary of Merck KGaA; and

WHEREAS, Licensee is engaged in the development, marketing and sale of pharmaceutical products in the Exclusive Territory (as hereinafter defined); and

WHEREAS, SONNET wishes to license to Licensee, on an exclusive basis, the right to research, Develop, import, export, market, use and Commercialize the Product (as hereinafter defined) in the DPN Field (as hereinafter defined) and to grant Licensee a non-exclusive option to acquire rights to the Product in the CIPN Field (as hereinafter defined) in the Exclusive Territory; and

WHEREAS, SONNET further wishes to grant Licensee a limited option to negotiate an expansion of the Exclusive Territory;

NOW, THEREFORE, in consideration of the promises and mutual covenants contained herein, the Parties agree to as follows:

ARTICLE 1 - DEFINITIONS

- **1.2** "Calendar Quarter" means each three- (3-) month period commencing January 1, April 1, July 1 or October 1 of any Calendar Year; provided, however, that (a) the first Calendar Quarter of the Term shall extend from the Effective Date to the end of the first full Calendar Quarter thereafter, and (b) the last Calendar Quarter of the Term shall end upon the expiration or termination of this Agreement.
- **"Calendar Year"** means the period beginning on the 1st of January and ending on the 3ft of December of the same year; *provided, however*, that (a) the first Calendar Year of the Term shall commence on the Effective Date and end on December 31 of the same year and (b) the last Calendar Year of the Term shall commence on January 1 of the Calendar Year in which this Agreement terminates or expires and end on the date of termination or expiration of this Agreement.
- **1.4 "Change of Control"** means, with respect to a Person: (a) a transaction or series of related transactions that results in the Sale or other disposition of more than 50% of such Person's assets; or (b) a merger or consolidation in which such Person is not the surviving corporation or in which, if such Person is the surviving corporation, the shareholders of such Person immediately prior to the consummation of such merger or consolidation do not, immediately after consummation of such merger or consolidation, possess, directly or indirectly through one or more intermediaries, a majority of the voting power of all of the surviving entity's outstanding stock and other securities and the power to elect a majority of the members of such Person's board of directors; but excluding transactions related to a capital increase of Licensee.
- 1.5 "CIPN" means chemotherapy-induced peripheral neuropathy.
- 1.6 "CIPN Field" means all prophylactic, palliative and therapeutic uses of the Product for CIPN in humans.
- 1.7 "Clinical Trial" means a clinical trial in human subjects that has been approved by a Regulatory Authority, designed to measure the safety and/or efficacy of the Product. Clinical Trials shall include Phase I Trials, Phase II Trials and Phase III Trials. For avoidance of doubt, the Product will be administered to patients [***].
- 1.8 "Collaboration" means the activities contemplated under this Agreement related to the Development of the Product for use in the Field.
- **"Combination Product"** means a Product, used in the context of the SONNET Patents, that: (a) includes one or more active ingredients in addition to the Compound; or (b) is combined with one or more products, devices, pieces of equipment or components.
- **1.10 "Commercialization" or "Commercialize"** means any and all activities undertaken before and after Regulatory Approval of a MAA for the Product and that relate to the marketing, promoting, distributing, importing or exporting for sale, offering for sale, and selling of the Product, and interacting with Regulatory Authorities regarding the foregoing.

-6-

- 1.11 "Commercially Reasonable Efforts" means: (a) with respect to the efforts to be expended by a Party with respect to any objective, such reasonable, diligent, and good faith efforts as such Party would normally use to accomplish a similar objective under similar circumstances; and (b) with respect to the Product, efforts and resources similar to those employed by companies in a similar stage of development and available resources as Licensee or SONNET to Develop, or Commercialize a product of similar market potential at a similar stage in its product life, taking into account, without limitation, commercial, legal and regulatory factors, target product profiles, product labeling, past performance, the regulatory environment and competitive market conditions in the therapeutic area, safety and efficacy of the Product, the strength of its proprietary position and such other factors as such companies may reasonably consider (including resource availability), all based on conditions then prevailing. For the avoidance of doubt, "Commercially Reasonable Efforts" with respect to the Product Development shall be assessed independently of Licensees' or SONNET's other activities that are not related to the Product Development and shall require what a diligent person would do to perform a sound and reasonable Development of the Product. For clarity, "Commercially Reasonable Efforts" will not mean that a Party guarantees that it will actually accomplish the applicable task or succeed in the targeted objective.
- 1.12 "Compound" means recombinant human interleukin-6 having the sequence set forth on Schedule 1.12, including any derivative, fragment or conjugate thereof.
- 1.13 "Confidential Information" of a Party means any information relating to the business, operations or products of a Party or any of its Affiliates, including any Know-How and biological or chemical materials not known or generally available to the public, that such Party discloses to the other Party under this Agreement, or otherwise becomes known to the other Party by virtue of this Agreement. For any Party, this Agreement and the terms and conditions herein are deemed "Confidential Information" of the other Party.
- **1.14 "Controlled"** means, with respect to (a) any Patent Right, (b) any Know-How or (c) any biological, chemical or physical material, that a Party or one of its Affiliates owns or has a license or sublicense to such Patent Right, Know-How or material (or in the case of material, has the right to physical possession of such material) and has the ability to grant a license or sublicense to, or to assign its right, title and interest in and to, such Patent Right, Know-How or material as provided for in this Agreement, without violating the terms of any agreement or other arrangement with any Third Party.
- **1.15** "Cover", "Covering" or "Covered" means, with respect to the Product, that the use, sale, or offer for sale of the Product would, except for a license granted under this Agreement, infringe a Valid Claim in the country in which the activity occurs.
- 1.16 "Development" or "Develop" means, with respect to the Product, the performance of all pre-clinical and clinical developments (including toxicology, pharmacology, test method development and stability testing, process development, formulation development, quality control development, statistical analysis), Clinical Trials (excluding Clinical Trials conducted after Regulatory Approval of an NDA), manufacturing and regulatory activities that are required to obtain Regulatory Approval of the Product in the Territory.
- 1.17 "DPN" means diabetic peripheral neuropathy.

-7.

- **1.18** "DPN Field" means all prophylactic, palliative and therapeutic uses of the Product for DPN in humans.
- 1.19 "Existing Third Party Agreement(s)" means the agreement(s) set forth on Schedule 1.19.

- **1.20 "Exclusive Territory"** means the following ASEAN countries (i) Malaysia, (ii) Singapore, (iii) Indonesia, (iv) Thailand, (v) Philippines, (vi) Vietnam (vii) Brunei, (viii) Myanmar, (ix) Lao PDR and (x) Cambodia.
- 1.21 "Expansion Territory" means the People's Republic of China, Hong Kong and India.
- 1.22 "FDA" means the United States Food and Drug Administration, or a successor federal agency thereto.
- 1.23 "Field" means the DPN Field and, if applicable, the CIPN Field. Field shall not include any other uses of the Product in humans.
- **1.24 "First Commercial Sale"** means, on a country-by-country basis and Product-by-Product basis, the first commercial transfer or disposition for value of a Product in such country to a Third Party by Licensee or any of its Affiliates or Sublicensees.
- 1.25 "Governmental Body" means any: (a) nation, principality, state, commonwealth, province, territory, county, municipality, district or other jurisdiction of any nature; (b) federal, state, local, municipal, foreign or other government; (c) governmental or quasi governmental authority of any nature (including any governmental division, subdivision, department, agency, bureau, branch, office, commission, council, board, instrumentality, officer, official, representative, organization, unit, body or entity and any court or other tribunal); (d) multi-national or supranational organization or body; or (e) individual, entity, or body exercising, or entitled to exercise, any executive, legislative, judicial, administrative, regulatory, police, military or taxing authority or power of any nature.
- 1.26 "IFRS" means the International Financial Reporting Standards, which are the set of accounting standards and interpretations and the framework in force on the Effective Date and adopted by the European Union as issued by the International Accounting Standards Board (IASB) and the International Financial Reporting Interpretations Committee (IFRIC), as such accounting standards may be amended from time to time.
- 1.27 "IND" means an investigational new drug application submitted to any applicable Regulatory Authorities for approval to commence Clinical Trials in a given jurisdiction.
- 1.28 "Know-How" means any scientific or technical information, results and data of any type whatsoever, in any tangible or intangible form whatsoever, that is not in the public domain or otherwise publicly known, including discoveries, inventions, trade secrets, devices, databases, practices, protocols, regulatory filings, methods, processes (including manufacturing processes, specification and techniques), techniques, concepts, ideas, specifications, formulations, formulae, data (including pharmacological, biological, clinical and analytical information, quality control, trial and stability data), case reports forms, medical records, data analyses, reports, studies and procedures, designs for experiments and tests and results of experimentation and testing (including results of research or development), summaries and information contained in submissions to and information from ethical committees, or Regulatory Authorities, and manufacturing process and development information, results and data, whether or not patentable, all to the extent not claimed or disclosed in a patent or patent application. The fact that an item is known to the public shall not be taken to exclude the possibility that a compilation including the item, and/or a development relating to the item, is (and remains) not known to the public. "Know-How" includes any rights including copyright, database or design rights protecting such Know-How. "Know-How" excludes Patent Rights.

-8-

- 1.29 "Law" or "Laws" means all applicable laws, statutes, rules, regulations, ordinances and other pronouncements having the binding effect of law of any Governmental Body.
- 1.30 "Liquidation Event" means any liquidation, dissolution, winding-up, or Change of Control of either Party, irrespective of its legal qualification.
- 1.31 "MAA" means a Marketing Authorization Application submitted pursuant to the requirements of the FDA, as more fully defined in 21 U.S. C.F.R. § 314.3 et seq., a Biologics License Application submitted pursuant to the requirements of the FDA, as more fully defined in 21 U.S. C.F.R. § 601, and any equivalent application submitted in any country in the Exclusive Territory, including all additions, deletions or supplements thereto, and as any and all such requirements may be amended, or supplanted, at any time.
- 1.32 "Net Margin" means Net Sales less New Life's actual cost of goods sold for the Product less New Life's sales and marketing expenses which sales and marketing expenses shall not exceed [***]% of Net Sales.
- 1.33 "Net Sales" means the gross amounts invoiced by Licensee or any of its Affiliates for sales of Product to independent or unaffiliated Third Party purchasers of such Product, less those deductions with respect to such sales that are either included in the billing as a line item as part of the gross amount invoiced, or otherwise documented as a deduction in accordance with IFRS to be specifically attributable to actual sales of such Product. Such deductions may also include any bad debt (provided that if such bad debt is subsequently collected it will be added to Net Sales).

If a Product under this Agreement is sold in the form of a Combination Product, then Net Sales for such Combination Product shall be determined on a country-by-country basis by mutual agreement of the Parties in good faith, taking into account the perceived relative value contributions of the Product and the other ingredient or component in the Combination Product, as reflected in their respective market prices. In case of disagreement, an independent expert designated by mutual agreement of both Parties or, failing such agreement, designated by the International Chamber of Commerce, shall determine such relative value contributions and such determination shall be final and binding upon the Parties.

9

In the event Product is "bundled" for sale together with one or more other products in a country (a "**Product Bundle**"), then Net Sales for such Product sold under such arrangement shall be determined on a country-by-country basis by mutual agreement of the Parties in good faith taking into account the relative value contributions of the Product and the other products in the Product Bundle, as reflected in their individual sales prices. In case of disagreement, an independent expert designated by mutual agreement of both Parties or, failing such agreement, the International Chamber of Commerce shall determine such relative value contributions and such determination shall be final and binding upon the Parties.

- **"Patent Right"** means: (a) an issued or granted patent, including any extension, supplemental protection certificate, registration, confirmation, reissue, reexamination, extension or renewal thereof; (b) a pending patent application, including any continuation, divisional, continuation-in-part, substitute or provisional application thereof; and (c) all counterparts or foreign equivalents of any of the foregoing issued by or filed in any country or other jurisdiction.
- 1.35 "Person" means any natural person, corporation, firm, business trust, joint venture, association, organization, partnership or other business entity, or any government or agency or political subdivision thereof.
- 1.36 "Phase I Trial" means a Clinical Trial in which the Product is administered to human subjects at multiple dose levels with the primary purpose of determining safety, metabolism, and pharmacokinetic and pharmacodynamics properties of the Product, and consistent with 21 CFR § 312.2l(a) or its equivalent in the applicable country in the Territory.

- 1.37 "Phase IB/IIA Trial" means a Clinical Trial in which the Product is administered to human subjects at multiple dose levels with the primary purposes of determining pharmacological or clinical activity (including dose response, dose escalation, duration of effect or kinetic/dynamic relationship assessments and to make a preliminary determination of efficacy and safety of the Product in the target patient population to permit the design of a Phase IIB or Phase III Trial as the case may be
- 1.38 "Phase II Trial" means a Clinical Trial of the Product in human patients, the principal purposes of which are to make a preliminary determination that the Product is safe for its intended use, to determine its optimal dose, and to obtain sufficient information about the Product's efficacy to permit the design of Phase III Trials, and consistent with 21 CFR 312.21(b) or its equivalent in the applicable country in the Territory.
- 1.39 "Phase III Trial" means a human Clinical Trial of the Product, which trial is designed (a) to establish that the Product is safe and efficacious for its intended use; (b) to define warnings, precautions and adverse reactions that are associated with the Product in the dosage range to be prescribed; and (c) consistent with 21 CFR § 312.21(c) or its equivalent in the applicable country of the Territory.
- 1.40 "Product" means any pharmaceutical product, including any formulation thereof, containing or comprising the Compound used in the frame of the SONNET Patents.

-10-

- **1.41** "Regulatory Authority" means (a) the FDA, or (b) any regulatory body with similar regulatory authority over pharmaceutical or biotechnology products in any other jurisdiction anywhere in the world including the Exclusive Territory.
- **1.42** "Regulatory Approval" shall mean the receipt from a Regulatory Authority by Licensee, its Affiliates, or Sublicensees, of approval to lawfully market a Product in the corresponding jurisdiction in the Exclusive Territory.
- 1.43 "Royalty Term" means, on a country-by-country and Product-by-Product basis in the Territory, the period from the First Commercial Sale of such Product in such country until the Commercialization of the Product by Licensee ceases in the DPN Field or, if applicable, the CIPN Field.
- 1.44 "Sale" means the sale and transfer of the entire outstanding and issued share capital of Licensee (100%) to any Third Party whatsoever in a single Transaction or series of related transactions.
- **1.45** "Section" means any section or Article of this Agreement.
- 1.46 "SONNET Data" means all existing pre-clinical and clinical data related to the Product in the DPN Field in possession or Control of SONNET as of the Effective Date.
- **1.47 "SONNET Know-How"** means all Know-How that relates to the Compound and that is Controlled by SONNET as of the Effective Date and is necessary in the research, Development, use, or Commercialization of the Product in the DPN Field.
- **"SONNET Patents"** means all Patent Rights set forth on Schedule 1.47 hereto, that are Controlled by SONNET as of the Effective Date and any additional Patents generated during the Term of this Agreement.

Notwithstanding the above, if SONNET decides to file any new patent Covering the same uses of the Compound in the DPN Field as the Patent Rights set forth in Schedule 1.47, then such new patent(s) shall, immediately upon filing, become an integral part of the SONNET Patents licensed to Licensee under Section 2.1. For clarity, if SONNET files any new patent covering the same uses as the SONNET Patents, but unrelated to the Compound, then such Patent Rights shall not fall within SONNET Patents within this Agreement.

- 1.49 "SONNET Technology" means the SONNET Know-How, the SONNET Data and the SONNET Patents.
- **1.50** "Sublicensee" means a Person other than an Affiliate of Licensee, to which Licensee (or its Affiliate) has granted sublicense rights pursuant to Section 2.2; for the sake of clarity, "Sublicensee" shall exclude distributors.
- 1.51 "Tax" or "Taxes" means any federal, state, local or foreign income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental, customs duties, capital stock, franchise, profits, withholding, social security, unemployment, disability, real property, personal property, sales, use, transfer, registration, value added, goods and services, alternative or add-on minimum, estimated, or other tax of any kind whatsoever imposed by any Governmental Body, including any interest, penalty, or addition thereto.

-11-

- **1.52** "**Territory**" collectively means the Exclusive Territory and if applicable, the Expansion Territory.
- 1.53 "Third Party" shall mean any Person other than a Party or an Affiliate of a Party.
- 1.54 "Valid Claim" means a claim of an issued and unexpired SONNET Patent, filed in the related country/-ies in the Territory, which has not lapsed or been revoked, abandoned or held unenforceable or invalid by a final decision of a court or governmental or supra-governmental agency of competent jurisdiction, unappealable or unappealed within the time allowed for appeal, and which has not been disclaimed, denied or admitted to be invalid or unenforceable through reissue, reexamination or disclaimer or otherwise.
- 1.55 Other Terms. The definition of each of the following terms is set forth in the Section of the Agreement indicated below:
 - "Action" has the meaning set forth in Section 6.5 (b).
 - "SONNET Indemnitees" has the meaning set forth in Section 9.1.
 - "SONNET Royalty Rate" has the meaning set forth in Section 5.3.
 - "Commercialization Plan" has the meaning set forth in Section 3.2.
 - "Controlling Party" has the meaning set forth in Section 6.6 (c).
 - "Development Plan" has the meaning set forth in Section 3.1.
 - "Executive Officers" has the meaning set forth in Section 11.2.
 - "Licensee Indemnitees" has the meaning set forth in Section 9.2.

"Term" has the meaning set forth in Section 10.1.

"Third Party Action" has the meaning set forth in Section 6.5 (a).

ARTICLE 2 - GRANT OF LICENSE AND OPTION

2.1 Grant of License. Subject to the terms and conditions of this Agreement, SONNET hereby grants to Licensee an exclusive, non-transferable, royalty-bearing right and license (with the right to sublicense subject to Section 2.2) under the SONNET Technology, to Develop, market, import, use and Commercialize the Product in the Field in the Exclusive Territory. For the avoidance of doubt, the license granted to Licensee hereunder shall not include the right to make, have made or export the Product outside the Exclusive Territory.

-12-

- **2.2 Grant of Sublicense by Licensee.** The Licensee shall have the right to grant sublicenses under the license granted in Section 2.1 with SONNET's prior written consent such consent not to be unreasonably withheld to its Affiliates, subsidiaries, sub-distributors or subcontractors for the purpose of conducting Development or Commercialization activities in the Exclusive Territory. The financial conditions for such sublicense shall reflect fair market value for the Product. The granting by Licensee of any sublicense shall not relieve Licensee of its obligations hereunder. Licensee shall provide SONNET with a copy of each such executed sublicense agreement; provided, that Licensee may redact from each such sublicense agreement all provisions, that are not relevant to Licensee's performance hereunder.
- 2.3 Non-Compete. Except as provided herein, SONNET hereby covenants not to practice, and not to permit or cause any of its Affiliates to develop, use, sell, have sold, offer for sale, or otherwise commercialize any Compound or Product for use in the DPN Field in the Exclusive Territory during the Term.
- 2.4 No Implied Licenses. Only those licenses expressly granted in this Agreement have effect. No license or other intellectual property interest or rights to the Compound or Product outside the DPN Field is granted by implication or any method that is not express. In addition, SONNET shall be deemed to retain such rights to the SONNET Technology as may be necessary or useful to the performance of SONNET's obligations hereunder.
- 2.5 Transfer. Upon receipt of a written request by Licensee, SONNET shall transfer to Licensee, at Licensee's cost and expense, all relevant SONNET Know-How and SONNET Data necessary for Licensee to perform its obligations hereunder. Such transfer shall take place in an orderly fashion and in manner such that the value, usefulness and confidentiality of the SONNET Know-How and SONNET Data are preserved in all material respects throughout their transfer.
- 2.6 Territory Expansion Option. SONNET hereby grants Licensee an exclusive option to negotiate the right to expand the Exclusive Territory to include one or more countries in the Expansion Territory starting on the Effective Date and automatically terminating on December 31, 2021 unless extended by mutual consent of the Parties (the "Option Period"). If the parties fail to reach agreement on the terms under which this License will be amended to include the Expansion Territory within the Option Period, the option shall expire and shall be of no further force or effect.
- 2.7 Non-Exclusive Option to CIPN Field. SONNET hereby grants Licensee a non-exclusive option during the Option Period to negotiate an expansion of the scope of the License to include the CIPN Field. If the parties fail to reach agreement on the terms under which this License will be amended to include the CIPN Field within the Option Period, the option to include the CIPN Field shall automatically expire and shall be of no further force or effect. For the avoidance of doubt, nothing herein shall prevent or preclude SONNET from negotiating or executing a license or other agreement for the CIPN Field with a Third Party during the Option Period.

-13-

2.8 Buy Back/Give Back Option.

(a) Licensee hereby grants SONNET an exclusive option to buy back the rights granted Licensee under this Section 2 for either or both the DPN Field or the CIPN Field (in the event the License is expanded to include the CIPN Field) for all or any country within the Exclusive Territory (the "Buy Back Option") prior to the initiation of a Phase III Trial for the Product (the "Buy Back Option Period"). During the Buy Back Option Period, SONNET shall provide a written notice to Licensee of its intention to exercise the Buy Back Option. If SONNET declines or otherwise fails to exercise the Buy Back Option prior to the expiration of the Buy Back Option Period, then SONNET'S Buy Back Option shall terminate and Licensee may continue to Develop and Commercialize such Products in the DPN Field and if applicable, the CIPN Field in the Exclusive Territory as provided herein.

(b) SONNET hereby grants Licensee an option to give back the rights granted Licensee under this Section 2 for either or both the DPN Field or the CIPN Field (in the event the License is expanded to include the CIPN Field) for all or any country within the Exclusive Territory (the "Give Back Option") which Give Back Option must be exercised during the Buy Back Option Period. Licensee shall provide a written notice to SONNET of its intention to exercise the Give Back Option. If Licensee declines or otherwise fails to exercise its Give Back Option prior to the expiration of the Buy Back Option Period, then Licensee's Give Back Option shall terminate and Licensee shall continue to Develop and Commercialize such Products in the DPN Field and if applicable, the CIPN Field in the Exclusive Territory as provided herein.

(c) In the event that SONNET elects to exercise its Buy Back Option as provided in Section 2.7(a) or Licensee elects to exercise the Give Back Option as provided in Section 2.7(b), then the Parties shall negotiate in good faith the terms under which the SONNET may buy back or Licensee shall give back the rights granted Licensee hereunder. In the event the Parties are unable to reach mutual agreement, then the Buy Back Option or Give Back Option shall expire and be of no further force or effect and Licensee shall retain the rights granted hereunder subject to the terms and conditions of this Agreement.

ARTICLE 3 - DEVELOPMENT AND COMMERCIALIZATION

3.1 Development of the Product by Licensee. Licensee shall have the exclusive right in the Exclusive Territory to research, Develop and Commercialize the Product and to conduct (either itself or through its Affiliates, agents, subcontractors and/or Sublicensees) all Clinical Trials and non-clinical studies Licensee (with the approval of the JDC) believes appropriate to obtain Regulatory Approval for the Product in the DPN Field. All costs associated with the development and Commercialization of the Product in the DPN Field in the Exclusive Territory shall be borne by Licensee. The Development of the Product shall be governed by a development plan, prepared and adopted by Licensee, that describes the proposed overall program of Development (the "Development Plan"), which Development Plan will be updated by Licensee at least twice annually.

-14-

3.2 Commercialization. Subject to the terms and conditions of this Agreement, Licensee shall have the exclusive right to Commercialize the Product itself or through one or more Third Parties selected by Licensee in the Exclusive Territory in the Field. The Commercialization of the Product shall be governed by a commercialization plan that describes the contemplated overall program of Commercialization (the "Commercialization Plan"), which Commercialization Plan shall be prepared and adopted by Licensee nine (9) months prior to the expected Commercialization of the Product. Such Commercialization Plan shall thereafter be updated by Licensee at least twice annually.

- 3.3 Manufacturing and Supply. Subject to the terms and conditions of this Agreement, SONNET shall manufacture the Compound and the Product itself or through one or more Third Parties selected by SONNET, and according to the Product development timeline set forth in Schedule 3.3 hereto. The Parties shall negotiate a supply agreement under which SONNET would supply the Compound and Products to Licensee for Development and Commercialization within the DPN Field and if applicable, the CIPN Field in the Exclusive Territory. The terms of such supply agreement shall be negotiated in good faith, it being understood and agreed that the Compound or Product would be sold to Licensee at a price equal to the then Standard Cost of the Compound or Product plus [***] percent ([****]%). The details of such Standard Cost of the Compound or Product would be made available prior to initiation of Clinical Trial.
- 3.4 Regulatory Filings. As between SONNET and Licensee and subject to Sections 2.7 and 10.5, Licensee shall own and maintain all regulatory filings and Regulatory Approvals for the Product in the Exclusive Territory and if applicable the Expansion Territory, including all INDs and MAAs.
- 3.5 Diligence. Licensee will use Commercially Reasonable Efforts to Develop and Commercialize the Product in each of the countries within the Exclusive Territory in the DPN Field and if applicable, the CIPN Field. For the avoidance of doubt, with respect to any pivotal Phase III Trials conducted in the Exclusive Territory, Licensee shall use Commercially Reasonable Efforts to Develop and Commercialize the Product. SONNET shall use commercially reasonable efforts to schedule and attend a pre-IND meeting with the FDA related to the Product within six (6) months of the Effective Date with the objective of securing a favorable outcome to allow commencement of Phase IB/IIA Clinical Trials for the Product.
- **3.6 Annual Reporting.** Licensee shall, on each anniversary of the Effective Date, provide SONNET with a written report summarizing in reasonable detail its Development and, as applicable, the Commercialization activities conducted during the preceding Calendar Year.
- 3.7 Trademarks. Licensee shall have the sole authority to create, select and register trademarks in the Territory for any Product subject to the consent of SONNET which consent shall not be unreasonably withheld and shall own all such trademarks.

_15.

ARTICLE 4 - DEVELOPMENT MANAGEMENT

- **4.1 Joint Development Committee.** As soon as practicable after the Effective Date, the Parties shall establish a committee to facilitate the Development of the Product (the "Joint Development Committee" or "JDC") as follows:
 - (a) Composition of the JDC. The Collaboration shall be conducted under the direction of a JDC comprised of two (2) representatives of Licensee and two (2) representatives of SONNET. Each Party shall appoint its respective representatives to the JDC from time to time, and may substitute one or more of its representatives, in its sole discretion, effective upon notice to the other Party of such change. Each Party shall have at least one JDC representative who is a senior employee (director level or above), and all JDC representatives shall have appropriate research, preclinical, manufacturing, clinical development or commercialization expertise and ongoing familiarity with the Collaboration. Additional representatives or consultants may from time to time, by mutual consent of the Parties, be invited to attend JDC meetings, subject to such representatives' and consultants' written agreement to comply with the requirements of Article 7. Each Party shall bear its own expenses relating to attendance at such meetings by its representatives.
 - (b) JDC Chairperson. The "JDC Chairperson" shall be a representative of SONNET. The JDC Chairperson's responsibilities shall include (a) scheduling meetings; (b) setting agendas for meetings with solicited input from other members; (c) confirming and delivering minutes to the JDC for review and final approval; and (d) conducting effective meetings, including ensuring that objectives for each meeting are set and achieved.
 - (c) Meetings. The JDC shall meet in accordance with a schedule established by mutual written agreement of the Parties and from time to time promptly upon the request of either Party, but no less frequently than twice per Contract Year, with the location for such meetings alternating between Licensee and SONNET facilities (or such other locations as are determined by the JDC). Alternatively, the JDC may meet by means of teleconference, videoconference or other similar communications equipment.
 - (d) JDC Responsibilities. The JDC shall have the following responsibilities with respect to the Collaboration:
 - (1) determining the overall Development strategy for the Collaboration;
 - (2) reviewing for approval (i) the annual update to the Development Plan and (ii) any modifications to such Development Plan in each case within thirty (30) days of each submission to the JDC;
 - (3) determining each Party's responsibilities under the Development Plan consistent with <u>Section 3.1</u>;

-16

- (4) facilitating the transfer of Know-How and Confidential Information between the Parties for purposes of conducting the Development Plan;
- (5) reviewing any new intellectual property filings and assessing the applicability of such patents to the Development Plan;
- (5) regularly assessing the progress of the Parties in their conduct of the Development Plan and against the timelines and budgets contained therein, reviewing relevant data, and considering issues of priority; and
- (6) performing such other activities as are contemplated under this Agreement and, subject to Section 12.10, that the Parties mutually agree shall be the responsibility of the JDC.
- 4.2 Appointment of Subcommittees, Project Teams and Collaboration Managers. The JDC shall be empowered to create such subcommittees of itself and additional project teams as it may deem appropriate or necessary. Each such subcommittee and project team shall report to the JDC, which shall have authority to approve or reject recommendations or actions proposed thereby subject to the terms of this Agreement. Each Party shall also designate a "Collaboration Manager." The Collaboration Managers will be responsible for the day-to-day coordination of the Collaboration and will serve to facilitate communication between the Parties. Each Party may change its designated Collaboration Manager from time to time upon written notice to the other Party.
- **Reports and Minutes.** Each Party will provide the members of the JDC with written copies of all materials they intend to present at the JDC meeting. The JDC may also request at any time specific data or information related to Development activities or that a written report be prepared in advance of any meeting summarizing certain material data and information arising out of the conduct of the Development activities and the Party or appropriate committee to whom such request is made shall promptly provide to the other Party or JDC such report, data or information. A secretary shall be appointed for each meeting and shall prepare minutes of the meeting, which shall provide a description in reasonable detail of the discussions held at the meeting and a list of any actions, decisions or determinations approved by the JDC.

4.4 Decision-Making and Dispute Resolution.

(a) <u>Voting</u>. With respect to decisions of the JDC, the representatives of each Party shall have collectively one vote on behalf of such Party. For each meeting of the JDC, at least two (2) representatives of each Party shall constitute a quorum. Action on any matter may be taken at a meeting, by teleconference, videoconference or by written agreement.

(b) <u>Decision-Making</u>. The JDC shall operate by consensus, subject to the dispute resolution process set forth in <u>Section 4.4.3</u> below.

-17-

- (c) <u>Dispute Resolution</u>. The JDC shall attempt to resolve any and all disputes relating to the Collaboration by unanimous consensus. In the event the JDC is unable to reach a unanimous consensus with respect to any such dispute, then the following dispute resolution provisions shall apply.
 - (1) With respect to any dispute over which the JDC has authority pursuant to Section 4.1(d), except those disputes related to the scope of the JDC's powers under Section 4.1(d)(6), Licensee shall have the final decision-making authority for the Development of the Product in the Exclusive Territory for use in the DPN Field following completion of the Phase II Trials except in the event that the proposed Development of the Product in the Exclusive Territory in SONNET's reasonable judgment creates an undue risk to the Development of the Product outside the Exclusive Territory then in such instance, SONNET shall have final decision making authority. For clarity, Commercialization activities within the Exclusive Territory shall not be the responsibility of the JDC and shall be subject to Licensee's final decision-making authority.
 - (2) With respect to all other disputes between the Parties regarding the interpretation, construction or application of this Agreement, the dispute shall be submitted to escalating levels of SONNET and Licensee senior management for review as described in this paragraph. If the dispute cannot be resolved despite escalation, then within thirty (30) days after the matter is referred to the JDC, the matter shall be referred to the Executive Officers to be resolved by negotiation in good faith as soon as is practicable but in no event later than thirty (30) days after referral. Such resolution, if any, by the Executive Officers shall be final and binding on the Parties. If the Executive Officers are unable to resolve such dispute within such thirty (30) day period, each Party will be free to pursue all rights available to it under law or equity as set forth in Section 11 hereof.
- **4.5 Dissolution of JDC.** The JDC shall be dissolved upon receipt of the first Regulatory Approval of the Product in the Exclusive Territory.

-18-

ARTICLE 5 - FINANCIALS

- 5.1 Upfront Payment. Within thirty (30) days of the Effective Date of this Agreement, in consideration of the rights and licenses granted herein, Licensee shall pay SONNET a non-fundable amount equal to five hundred thousand US dollars (\$500,000).
- **5.2 Deferred License Fee.** Licensee shall pay SONNET an additional non-refundable amount of \$1,000,000 within [***] ([***]) days of achievement of Milestone #[***] below.

5.3 Milestones.

(a) <u>Milestones</u>. Upon the first achievement of each of the milestone events set forth in the table below by Licensee or any Affiliate or Sublicensee, Licensee shall provide SONNET with written notice of such achievement and shall pay to SONNET the corresponding one-time milestone payment set forth below (each a "**Milestone**"):

1. []	Milestone Event	 Milestone Payment	
		<u> </u>	[***]
	2. [***]	\$ S	\$[***]

[***

5.4 Royalty; Monetary Third Party Obligations

(a) Royalty Rate. As consideration for the granting of rights to the SONNET Technology, Licensee shall, during the Royalty Term, pay to SONNET a royalty on the Net Sales of the Product as set forth below:

Annual Net Sales	Royalty on Net Sales
\$ [***]-\$[***]	12%
\$ [***]-\$[***]	[***]%
\$ [***]-\$[***]	[***]%
>\$[***]	30%

(b) Except as expressly set forth in this Agreement, such royalty shall not be subject to any offset or reduction for any reason, including but not limited to any Taxes, royalties, milestone payments or other consideration that Licensee may pay to any Third Party.

(c) Considerations under Existing Third Party Agreements. SONNET shall be obliged to and be responsible for paying all monetary obligations owed by SONNET or any of its Affiliates to Third Parties (the "Monetary Third Party Obligations") for the Product, including without limitation royalty and other payment obligations, under Existing Third Party Agreements.

-19-

(d) Sublicensing Receipts. Licensee shall pay SONNET [***]% of all revenue received through sub-licensing of each Product (including without limitation upfront, milestone payments, annual maintenance fees and similar payments, but excluding payments directly related to research "full-time equivalents". In the event the sublicensing of the Product results in a Liquidation Event, then SONNET shall be entitled to receive [***]% of the proceeds received by Licensee.

(e) Royalty and Sublicensing Fee Reporting Currency Conversion. Commencing with the Calendar Quarter in which the First Commercial Sale of a Product is made by the Licensee or its Affiliate or any Sublicensee, Licensee shall submit to SONNET with each royalty payment, including any due royalty and Sublicensing Fee a report detailing its computation of (i) royalties for the SONNET Royalty Rate due on Net Sales and (ii) of the Sublicensing Fees, for the

corresponding Calendar Quarter. Such Report and the associated payments shall be due within [***] ([***]) days after the end of each Calendar Quarter All payments to SONNET hereunder shall be made in US dollars in the requisite amount to such bank account as SONNET may from time to time designate by written notice to Licensee. With respect to sales not denominated in Dollars, any amounts owed to SONNET by Licensee shall first be calculated in the currency of sale, and then such amounts shall be converted into US Dollars using the average of the month end daily currency exchange rates published by Bloomberg (or its successors) on the last day of the Calendar Quarter to which the report relates. The Parties may vary the method of payment set forth herein at any time upon mutual agreement, and any change shall be consistent with the local law at the place of payment or remittance.

(f) Record Retention, Inspection. Licensee shall keep or cause its Affiliates and Sublicensees to keep complete and accurate records in sufficient detail to enable Net Sales, royalties and Sublicensing Fees payable under this Section 5.4, to be established for a period of sixty (60) months after the date that such amounts were payable. Such records shall be consistent with Licensee's normal accounting principles. At the request of SONNET (but not more frequently than once each Calendar Year) an independent chartered or certified public accountant chosen and paid by SONNET but approved by the Licensee (which approval shall not be unreasonably withheld or delayed) shall be allowed access during ordinary business hours to such records pertaining to the preceding two (2) Calendar Years solely to verify the accuracy of any payments made to SONNET under this Section 5.4. The accountant shall not disclose to SONNET any information other than that which should properly be contained in a report of matters relevant to Net Sales and royalty calculation and, as the case may be, Sublicensing Fees' payment arising under this Section 5.4. As regards the Sublicensing Fees, Licensee shall cause its Sublicensees to make their records available to SONNET.

(g) Conversion of Royalties to Equity. In the event Licensee (i) files for an Initial Public Offering or (ii) is subject to a Change of Control, the Parties may mutually agree (1) to have the royalty stream for the Products convert to equity in such amount as agreed to between SONNET and Licensee, or (2) to require the successor to Licensee assume the royalty obligations due SONNET hereunder.

-20-

ARTICLE 6 - INVENTIONS, PATENTS AND CLINICAL DATA

- 6.1 Certification Under Drug Price Competition and Patent Restoration Act. Each Party shall immediately give written notice to the other Party of any certification filed pursuant to 21 U.S.C. Section 355(b)(2)(A) (or any amendment or successor statute thereto), of which it becomes aware and which claims either that any SONNET Patent, any Product or the Development, manufacture, use or Commercialization, of each of the foregoing, are invalid or unenforceable, or that no infringement will arise from the Development, manufacture, use or Commercialization of any similar product by a Third Party.
- **6.2 Listing of Patents.** SONNET shall have the sole right to determine which of the SONNET Patents, if any, shall be listed for inclusion in the Approved Drug Products with "Therapeutic Equivalence Evaluations" pursuant to 21 U.S.C. Section 355, or any successor law in the United States, together with any comparable laws or regulations in any other country in the Exclusive Territory.
- 6.3 Title to Inventions. SONNET is and shall at all times be the sole and exclusive owner of all right, title and interest in and to the SONNET Technology, other than Joint Inventions and Joint Patent Rights. A Party shall have and retain all right, title and interest in any Invention made solely by one or more employees or agents of such Party and or its Affiliates or other persons acting under its authority. The Parties shall jointly own rights in any Invention made jointly by one or more employees or agents of each Party and/or such Party's Affiliates or other persons acting under its authority ("Joint Inventions") and Patent Rights therein ("Joint Patent Rights"). For clarity, Inventions developed exclusively by one Party and such Party's Affiliates shall not be considered Joint Inventions. Subject to the rights and licenses granted under this Agreement, each Party shall have the right to practice and use, and grant licenses to practice and use, any Joint Inventions and Joint Patent Rights without the other Party's consent and has no duty to account to the other Party for such practice, use or license, and each Party hereby waives any right it may have under the laws of any country to require any such consent or accounting; provided however that in the event SONNET exercises its Buy Back Right or this Agreement is terminated by SONNET pursuant to Section 10.3 hereof, SONNET shall be automatically granted an exclusive license (even as to Licensee) in such Joint Patent Rights. Each Party shall be liable with respect to its own employees for compliance with any applicable legislation and its own policies concerning employee inventions, including payment of employee invention awards (if any).

6.4 Patent Prosecution and Maintenance.

(a) Licensee. Licensee shall have the right to file, prosecute and maintain each and all patents it owns Licensee shall bear all costs and expenses of filing, prosecuting and maintaining such patents.

-21-

(b) SONNET Patents. SONNET shall have the first right, and the obligation, to file, prosecute and maintain each and all SONNET Patents within the Exclusive Territory. Licensee's shall reimburse SONNET for all costs and expenses of filing, prosecuting and maintaining the SONNET Patents in the Exclusive Territory, subject to Section 6.4 (c). SONNET shall keep Licensee informed about the course of the filing and prosecution of SONNET Patents or related proceedings (e.g., interferences, oppositions, reexaminations, reissues, revocations or nullification s) in the Exclusive Territory in a timely manner, and to take into consideration the advice and recommendations of Licensee. At SONNET's request, Licensee will provide SONNET with reasonable assistance in prosecuting SONNET Patents to the extent possible, in particular by providing to SONNET any data related to the SONNET Patents which is under Licensee's Control and which is, in SONNET's reasonable judgment, needed to support the prosecution of any SONNET Patent; provided, however, that SONNET shall reimburse Licensee for Licensee's out-of-pocket expenses incurred in providing such assistance.

(c) Joint Patent Rights. SONNET shall have the first right, but not the obligation, to prepare, file, prosecute and maintain all Joint Patent Rights, by counsel of SONNET's choice. The cost of prosecution of said Joint Patent Rights shall be shared equally by the Parties. SONNET shall keep Licensee reasonably informed of progress with regard to the preparation, filing, prosecution and maintenance of the Joint Patent Rights, and shall provide to Licensee copies of all material patent office submissions within a reasonable amount of time following submission thereof by SONNET. In the event that SONNET desires to abandon or cease prosecution or maintenance of any Joint Patent Right, SONNET shall provide written notice to Licensee of such intention to abandon promptly after SONNET makes such determination, which notice shall be given no later than thirty (30) days prior to the next deadline for any action that must be taken with respect to such Joint Patent Right in the relevant patent office. In such case, Licensee shall have the right, in its discretion, exercisable upon written notice to SONNET delivered no later than fifteen (15) days after receipt of notice from SONNET, to assume responsibility for prosecution and maintenance of such Joint Patent Right, at its sole cost and expense and by counsel of its own choice.

(d) Patent Report. On each anniversary of the Effective Date during the Term and until such time that there is no longer a Valid Claim under a SONNET Patent, SONNET shall provide to Licensee an annual report reflecting the status of the SONNET Patents in the Exclusive Territory.

(e) Election not to file and prosecute SONNET Patents in the Exclusive Territory. If SONNET elects not to file a SONNET Patent, or to prosecute or maintain any existing SONNET Patent, in a country in the Exclusive Territory, then SONNET undertakes to notify Licensee in writing at least ninety (90) days before any deadline applicable to the filing, prosecution or maintenance of such SONNET Patent, as the case may be, or any other date by which an action must be taken to establish or preserve such SONNET Patent in such country of the Exclusive Territory. In such case, Licensee shall have the right, but not the obligation, to initiate the filing or to support the continued prosecution or maintenance of such SONNET Patent in the related country in the Exclusive Territory. Licensee shall notify SONNET of its decision, and SONNET shall reasonably cooperate with Licensee in this regard, including, if requested by Licensee, by assigning to Licensee all its right, title and interest in and to any such SONNET Patent in such country of the Exclusive Territory, with a view to transfer and assign the ownership of such SONNET Patent

For the sake of clarity, it is specified that no royalty whatsoever under Section 5.3 shall be due anymore to SONNET for the use of any such transferred SONNET Patent and no royalty or any other sort of indemnity shall be due to SONNET for the filing of new patents by Licensee in any unregistered country of the Exclusive Territory.

(f) Patent Term Extension. SONNET shall be responsible for obtaining patent term extensions wherever available for SONNET Patents Licensee shall provide SONNET with all relevant information, documentation and assistance in this respect. Any such assistance, supply of information and consultation shall be provided promptly and in a manner that will ensure that all patent term extensions for Products are obtained wherever legally permissible, and to the maximum extent available. In the event that any election with respect to obtaining patent term extensions is to be made, Licensee shall have the right to make such elections, and SONNET shall abide by all such elections.

6.5 Enforcement of Patents.

- (a) Notice. If either Party believes that any SONNET Patent is being infringed by a Third Party or if a Third Party claims that any SONNET Patent is invalid or unenforceable within the Exclusive Territory, the Party possessing such knowledge or belief shall notify the other Party and provide it with details of such infringement or claim that are known by such Party.
- (b) Right to Bring an Action. As long as it owns the relevant SONNET Patent, SONNET shall have the exclusive right to attempt to resolve such infringement or claim, including by filing an infringement suit, defending against such claim or taking other similar action (each, an "Action") and to compromise or settle such infringement or claim. If SONNET does not intend to take an Action, SONNET shall promptly inform Licensee which therefore is granted the right to initiate such an Action.

Notwithstanding the foregoing, each Party shall have the right to join an Action relating to a SONNET Patent, taken by the other Party at its own expense.

- (c) Costs of an Action. Subject to the respective indemnification obligations set forth in Section 9, the Party taking an Action under Section 6.5(b) shall assume all costs associated with such Action, including any possible assistance as detailed under Section 6.5 (e), to the exception of the expenses that the other Party may incur if it elects to join such Action.
- (d) Settlement. Neither Party shall settle or otherwise compromise any Action without the other Party's prior written consent. The settlement will be treated in accordance with the law of the country to which the settlement relates.
- (e) Reasonable Assistance. The Party who does not join an Action shall provide reasonable assistance to the other Party, including providing access to relevant documents and other evidence and making its employees available, subject to the other Party's reimbursement of any out-of-pocket expenses incurred by such assistance.

-23-

- **(f) Distribution of Amounts Recovered.** Any amounts recovered by the Party taking an Action pursuant to this Section 6.5, whether by settlement or judgment, shall be allocated in the following order:
- (i) to reimburse the Party taking such Action for any costs incurred;
- (ii) to reimburse the Party not taking such Action for its costs incurred in such Action, if it joins such Action; and
- (iii) the remaining amount of such recovery shall be retained by the Party taking the Action.

6.6 Third Party Actions Claiming Infringement.

- (a) Notice. If a Party becomes aware of any claim or action by a Third Party against either Party that claims that the Product, or its use, Development, manufacture or Commercialization infringes such Third Party's intellectual property rights (each, a "Third Party Action") in the Exclusive Territory, such Party shall promptly notify the other Party of all details regarding such claim or action that is reasonably available to such Party.
- (b) Right to Defend. SONNET shall have a first right, but not the obligation, to defend, at its sole expense, a Third Party Action If SONNET declines or fails to assert its intention to defend such Third Party Action within a brief time period (i.e. with sufficient time for Licensee to take whatever action may be necessary prior to the date on which such right to defend shall lapse), then Licensee shall have the right to defend such Third Party Action. The Party defending such Third Party Action shall have the sole and exclusive right to select its own counsel for such Third Party Action.
- (c) Consultation. The Party defending a Third Party Action pursuant to this Section 6.6 (the "Controlling Party") shall consult with the non-Controlling Party on all material aspects of the defense. The non-Controlling Party shall have a reasonable opportunity for meaningful participation in decision-making and formulation of defense strategy. The Parties shall reasonably cooperate with each other in all such actions or proceedings.
- (d) Appeal. In the event that a judgment in a Third Party Action is entered against the Controlling Party and an appeal is available, the Controlling Party shall have the first right, but not the obligation, to file such appeal. In the event the Controlling Party does not desire to file such an appeal, it will promptly, in a reasonable time period (i.e., with sufficient time for the nonControlling Party to take whatever action may be necessary) prior to the date on which such right to appeal will lapse or otherwise diminish, permit the non-Controlling Party to pursue such appeal at such non-Controlling Party's own cost and expense. If applicable law requires the other Party's involvement in an appeal, the other Party shall be a nominal party of the appeal and shall provide reasonable cooperation to such Party at such Party's expense.

-24-

- (e) Costs of an Action. Subject to the respective indemnification obligations of the Parties set forth in Article 9, the Controlling Party shall pay all costs associated with such Third Party Action other than the expenses of the other Party if the other Party elects to join such Action. Each Party shall have the right to join a Third Party Action defended by the other Party, at its own expense.
- (f) No Settlement Without Consent. No Controlling Party shall settle or otherwise compromise any Third Party Action by admitting that any SONNET Patent is invalid or unenforceable without the non-Controlling Party's prior written consent.

6.7 Clinical Data. All clinical data resulting from human clinical studies in the DPN Field as conducted by Licensee shall be jointly owned. Licensee shall provide all such clinical data to SONNET.

ARTICLE 7 – CONFIDENTIALITY

- 7.1 Confidentiality Obligations. Each Party agrees that in addition to and not in lieu of the confidentiality obligations set forth in the Mutual Confidentiality Agreement between the Parties dated [***], for the Term and for five (5) years thereafter, it shall, and shall ensure that its officers, directors, employees and agents shall, keep completely confidential and not publish or otherwise disclose and not use for any purpose, except as expressly permitted hereunder, any Confidential Information disclosed to it by the other Party pursuant to this Agreement. The foregoing obligations shall not apply to any Confidential Information disclosed by a Party hereunder to the extent that the receiving Party can demonstrate that such Confidential Information:
 - (i) was already known to the receiving Party or its Affiliates, other than under an obligation of confidentiality, at the time of disclosure;
 - (ii) was generally available to the public or otherwise part of the public domain at the time of its disclosure to the receiving Party;
 - (iii) became generally available to the public or otherwise part of the public domain after its disclosure and other than through any act or omission of the receiving Party in breach of this Agreement;
 - (iv) was subsequently lawfully disclosed to the receiving Party or its Affiliates by a Third Party without an obligation of confidentiality other than in contravention of a confidentiality obligation of such Third Party to the disclosing Party; or
 - (v) was developed or discovered by employees or agents of the receiving Party or its Affiliates who had no access to the Confidential Information of the disclosing Party.

Notwithstanding the above, a Party may disclose information to the extent that such disclosure is reasonably necessary in connection with:

- (vi) filing new patent applications or prosecuting or maintaining SONNET Patents, in accordance with the terms and conditions of this Agreement;
- (vii) prosecuting or defending any litigation;

25-

- (viii) conducting pre-clinical studies or Clinical Trials;
- (ix) seeking Regulatory Approval of the Product;
- (x) seeking additional equity investments, provided that only such information is disclosed that is directly related to the Product; or
- (xi) complying with any applicable law, including securities law and the rules of any securities exchange or market on which a Party's securities are listed or traded

In addition, in connection with any permitted filing by either Party of this Agreement with any Governmental Body, included but not limited to the U.S. Securities and Exchange Commission Agreement, the filing Party shall endeavor to obtain confidential treatment of economic, trade secret information and such other information as may be requested by the other Party, and shall provide the other Party with the proposed confidential treatment request with reasonable time for such other Party to provide comments, and shall include in such confidential treatment request all reasonable comments of the other Party disclosures set forth in clauses (i) through (v) above, the disclosing Party shall, where reasonably practicable, give such advance notice to the other Party of such disclosure requirement as is reasonable under the circumstances and will use its reasonable efforts to cooperate with the other Party in order to secure confidential treatment of such Confidential Information required to be disclosed.

- 7.2 **Publications.** Licensee shall not publish any information relating to the Product without the written consent of SONNET. If such information has already been publicly disclosed either prior to the Effective Date or after the Effective Date through no fault of Licensee or otherwise not in violation of this Agreement, Licensee shall provide SONNET with written notice prior to publication in a journal in which a submission is made by Licensee. In any case, Licensee shall submit to SONNET's written approval (which approval shall be granted or denied in SONNET's sole discretion) any publication or presentation (including, without limitation, in any seminars, symposia or otherwise) of information related directly or indirectly to the Product for review and approval.
- 7.3 Press Releases and Disclosure. It is understood that SONNET intends to issue a press release announcing the execution of this Agreement at a mutually agreed upon time and content (as attached hereto as Exhibit A) and that each Party thereafter may desire or be required to issue subsequent press releases relating to the Agreement or activities thereunder. Except as otherwise provided in this Section 7.3, neither Party may issue a press release relating to this Agreement or activities hereunder without the prior consent of the other Party (which consent shall not be unreasonably withheld, conditioned or delayed) and without complying with this Section 7.3; provided, however, that either Party may issue such press releases as it determines, based on advice of counsel, are strictly necessary to comply with laws or regulations or for appropriate market disclosure. If a Party wishes to issue a press release, it shall provide the other Party with a draft of such press release so that the other Party shall have sufficient time to review such release. If no comments are provided by the end of such a five (5) business day period following the receipt of the draft, the release will be deemed to have been approved by the other Party. Following the initial press release announcing this Agreement, either Party shall be free to disclose, without the other Party's prior written consent, the existence of this Agreement, the identity of the other Party and those terms of the Agreement which have already been publicly disclosed in accordance herewith.

-26-

ARTICLE 8 - REPRESENTATIONS AND WARRANTIES

- **8.1** Representations and Warranties. Each Party represents and warrants to the other Party that, as of the Effective Date:
 - (1) such Party is duly organized and validly existing under the Laws of the jurisdiction of its incorporation or organization;
 - (2) such Party has taken all action necessary to authorize the execution and delivery of this Agreement and the performance of each and all its obligations thereunder;
 - (3) this Agreement is a legal and valid obligation of such Party, binding upon such Party and enforceable against such Party in accordance with the terms of this Agreement and applicable Law. The execution, delivery and performance of this Agreement by such Party does not conflict with, breach or create in any Third Party, the right to accelerate, terminate or modify any agreement or instrument to which such Party is a party or by which such Party is bound, and does not violate any Law of any Governmental Body having authority over such Party; and such Party has all right, power and authority to enter into this Agreement, to perform its obligations under this Agreement.

In addition to the above, SONNET expressly represents and warrants to Licensee that it is in Control of, or is validly entitled to engage, each and all elements of the SONNET Technology transferred and/or licensed to Licensee, pursuant to this Agreement. SONNET further covenants and agrees that it shall promptly provide a copy of the minutes of the scheduled pre-IND meeting related to the Product with the FDA which FDA meeting minutes shall be deemed Confidential Information of SONNET.

8.2 No Further Representations and Warranties. SONNET gives no representations or warranties that the SONNET Patents which are patent applications will be granted or, if granted, neither that they will be valid nor that the exercise of the rights granted to Licensee hereunder will not infringe other patent rights or intellectual property rights vested in any Third Party. SONNET DISCLAIMS ALL OTHER WARRANTIES EXPRESS OR IMPLIED, INCLUDING WITHOUT LIMITATION, WARRANTIES TO TITLE OR NON-INFRINGEMENT OR IMPLIED WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE.

-27-

ARTICLE 9 - INDEMNIFICATION AND INSURANCE

- 9.1 Indemnification by Licensee. Licensee shall indemnify, defend and hold SONNET and its Affiliates and each of their respective employees, officers, directors and agents (the "SONNET Indemnitees") harmless from and against any and all liability, damage, loss, cost or expense (including reasonable attorneys' fees) to the extent arising out of Third Party claims or suits related to: (a) Licensee's negligence or willful misconduct; (b) Licensee's breach of its obligations under this Agreement; (c) breach by Licensee of its representations or warranties set forth in Section 8.1; (d) the Development, manufacture and Commercialization of Products, including without limitation product liability claims; provided, however, that Licensee's obligations pursuant to this Section 9.1 shall not apply (i) to the extent such claims or suits result from the negligence or willful misconduct of any of the SONNET Indemnitees, or (ii) with respect to claims or suits arising out of breach by SONNET of its representations, warranties or covenants set forth in Section 8.1.
- 9.2 Indemnification by SONNET. SONNET shall indemnify, defend and hold Licensee and its Affiliates and each of their respective agents, employees, officers and directors (the "Licensee Indemnitees") harmless from and against any and all liability, damage, loss, cost or expense (including reasonable attorney's fees) to the extent arising out of Third Party claims or suits (including Third Party Actions) related to: (a) SONNET's negligence or willful misconduct; (b) SONNET's breach of its obligations under this Agreement; or (c) breach by SONNET of its representations, warranties or covenants set forth in Section 8.1; provided, however, that SONNET's obligations pursuant to this Section 9.2 shall not apply (i) to the extent that such claims or suits result from the negligence or willful misconduct of any of Licensee Indemnitees or (ii) with respect to claims or suits arising out of a breach by Licensee of its representations or warranties set forth in Section 8.1.
- 9.3 No Consequential Damages. EXCEPT WITH RESPECT TO EACH PARTY'S INDEMNIFICATION OBLIGATIONS UNDER SECTION 9.1 OR SECTION 9.2, AS APPLICABLE, IN NO EVENT SHALL EITHER PARTY OR ANY OF ITS AFFILIATES BE LIABLE TO THE OTHER PARTY OR ANY OF ITS AFFILIATES FOR SPECIAL, INDIRECT, INCIDENTAL, CONSEQUENTIAL OR PUNITIVE DAMAGES, INCLUDING LOSS OF PROFITS, WHETHER IN CONTRACT, WARRANTY, TORT, NEGLIGENCE, STRICT LIABILITY OR OTHERWISE ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE TRANSACTIONS CONTEMPLATED HEREIN OR ANY BREACH HEREOF. NOTWITHSTANDING THE FOREGOING, NOTHING IN THIS AGREEMENT SHALL LIMIT EITHER PARTY FROM SEEKING OR OBTAINING ANY REMEDY AVAILABLE UNDER LAW FOR ANY BREACH OF BY THE OTHER PARTY OF ITS CONFIDENTIALITY AND NON-USE OBLIGATIONS UNDER ARTICLE 7.

-28-

- 9.4 Notification of Claims; Conditions to Indemnification Obligations. Except for the specifics foreseen under the scope of Section 6.6, as a condition to a Party's right to receive indemnification under this Article 9, it shall: (a) promptly notify the other Party as soon as it becomes aware of a claim or suit for which indemnification may be sought pursuant hereto; (b) cooperate, and cause the individual indemnites to cooperate, with the indemnifying Party in the defense, settlement or compromise of such claim or suit; and (c) permit the indemnifying Party to control the defense, settlement or compromise of such claim or suit, including the right to select defense counsel. In no event, however, may the indemnifying Party compromise or settle any claim or suit in a manner which admits fault or negligence on the part of the indemnified Party or any indemnitee without the prior written consent of the indemnified Party. Each Party shall reasonably cooperate with the other Party and its counsel in the course of the defense of any such suit, claim or demand, such cooperation to include without limitation using reasonable efforts to provide or make available documents, information and witnesses. The indemnifying Party shall have no liability under this Article 8 with respect to claims or suits settled or compromised without its prior written consent.
- 9.5 Insurance. During the Term, each Party shall obtain and maintain, at its sole cost and expense, insurance (including any self-insured arrangements) in types and amounts, that are reasonable and customary in the pharmaceutical and biotechnology industry for companies engaged in comparable activities. It is understood and agreed that this insurance shall not be construed to limit either Party's liability with respect to its indemnification obligations hereunder. Each Party will, except to the extent self-insured, provide to the other Party upon request a certificate evidencing the insurance such Party is required to obtain and keep in force under this Section 9.

ARTICLE 10 - TERM AND TERMINATION

- 10.1 Term of Agreement. The term of this Agreement (the "Term") shall commence on the Effective Date and unless earlier terminated as provided in this Section 10, shall continue in full force and effect, on a country-by-country and Product-by-Product basis, until the earlier of (i) the expiry of the Royalty Term in such country, or (ii) exercise by SONNET of the Buy Back Right set forth in Section 2.7 at which time this Agreement shall expire in its entirety in the case of termination under 10.1(ii) above or in all other cases with respect to such Product in such country.
- 10.2 Termination of the Agreement for Failure to Use Commercially Reasonable Efforts. SONNET may terminate this Agreement in its entirety, or on a country-by-country basis, upon sixty (60) days prior written notice to Licensee if Licensee fails to exercise Commercially Reasonable Efforts as required by Section 3.5 hereof.
- 10.3 Termination for Breach. Either Party may terminate this Agreement, and the rights and licenses granted hereunder, with a sixty (60) days prior notice to the other Party if the other Party breaches any material provision of this Agreement or the Supply Agreement to be entered into by the Parties, including any financial obligations under Sections 5.1, 5.2, 5.3 or 5.4, unless the other Party cures such breach within the period of such notice. Such termination shall be in addition to any other remedies available to the terminating Party at Law.
- 10.4 Termination for Bankruptcy. To the extent permitted by applicable Law, all rights and licenses granted by SONNET to Licensee pursuant to this Agreement shall automatically terminate on the insolvency or bankruptcy of such Party or its Affiliates, and each Party hereby claims the benefit of any applicable Law which may enable it to prevent such termination.

-29.

10.5 Effects of Termination.

- (b) Surviving Provisions. Sections 5, 7, 9, 10.5, 11 and 12 shall survive any termination of this Agreement.
- (c) Consequences of Termination. Upon any termination of this Agreement:
- (i) All licenses granted to Licensee under Section 2.1 shall terminate;
- (ii) At SONNET's sole discretion, SONNET has the right to assume legal responsibility for any Clinical Trials of the Product in the Exclusive Territory. Licensee shall, upon written request by SONNET, transfer to SONNET all regulatory documentation and Regulatory Approvals prepared or obtained by or on behalf of Licensee prior to the date of such termination, to the extent solely related to Products and transferable. In the case of SONNET's termination of this Agreement pursuant to Sections 10.2 or 10.3, the above transfer shall be at Licensee's cost and expense. In the case of Licensee's termination for breach the transfer shall be at SONNET's cost and expense.
- (iii) Licensee shall return to SONNET all relevant records and materials in its possession or Control containing or comprising the SONNET Know-How or such other Confidential Information of SONNET.
- (iv) Licensee shall, at SONNET's option, transfer to SONNET free of charge, all chemical, biological or physical materials relating to or comprising the Products, including clinical supplies of Products, that are owned or Controlled by Licensee, upon commercial terms to be mutually agreed upon between the Parties in good faith.
- (v) To the extent not prohibited by Law, Licensee shall wind down any ongoing Clinical Trials with respect to the Product, or at SONNET's option, transfer such clinical trials to SONNET, in which case Licensee shall provide SONNET with the relevant Clinical Trial supplies of the Product free of charge. In the case of SONNET's termination pursuant to Sections 10.2 or 10.3 the above transfer shall be at Licensee's cost and expense. In the case of Licensee's termination for breach the transfer shall be at SONNET's cost and expense.

-30-

- (vi) Licensee, its Affiliates and Sublicensees shall be entitled, during the six (6) month period following such termination, to sell any commercial inventory of Products which remains on hand as of the date of the termination, so long as Licensee pays to SONNET the royalties applicable to said subsequent sales in accordance with the terms and conditions set forth in this Agreement. Any commercial inventory remaining following such six (6) month period shall be offered for sale to SONNET at cost or otherwise destroyed.
- (vii) Licensee shall, if the termination is exercised by SONNET at SONNET's option, assign the trademarks owned by Licensee relating to the Product(s) to SONNET or otherwise transfer rights to such trademarks to SONNET.
- (d) Upon termination of this Agreement of any sort, each Sublicensee shall continue to have the rights and license set forth in its sublicense agreements, which agreements shall be automatically assigned to SONNET; provided, however, that such Sublicensee is not then in breach of any of its material obligations under its sublicense agreement.

ARTICLE 11 - DISPUTE RESOLUTION - JURISDICTION

- 11.1 Disputes. The Parties agree to first establish and follow procedures to facilitate the resolution of disputes arising out of or in relation with this Agreement in an expedient manner by mutual cooperation and without resort to litigation. In the event that the Parties are unable to resolve such dispute through diligent review and deliberation by their respective senior executives within thirty (30) days from the day that one Party had notified the issue as a dispute in written notice to the other Party, then either Party shall have the right to escalate such matter to their respective Executive Officers as further detailed under Section 11.2.
- 11.2 Escalation to Executive Officers. Either Party may, by written notice to the other Party, request that a dispute arising out of or in relation with this Agreement that remains unresolved by the respective senior executives of the Parties for a period of thirty (30) days be resolved by the Executive Officers, within fifteen (15) days after referral of such dispute to them. If the Executive Officers cannot resolve such dispute within fifteen (15) days after referral of such dispute to them, then, at any time after such fifteen (15) day period, either Party may proceed to enforce any and all of its rights with respect to such dispute.

For the purposes of Section 11.2, "Executive Officers" means, together, a member of the senior management of SONNET and the Chief Executive Officer of Licensee.

-31-

- Dispute Resolution. The Parties hereby agree that they will attempt in good faith to resolve any controversy or claim arising out of or relating to this Agreement 11.3 promptly by negotiations as provided above. Any dispute, controversy or claim initiated by either party arising out of, resulting from or relating to this Agreement, or the performance by either party of its obligations under this Agreement (other than bona fide Third Party actions or proceedings filed or instituted in an action or proceeding by a Third Party against a party), whether before or after termination of this Agreement, shall be settled by binding arbitration by submitting the same for arbitration pursuant to the rules of the International Chamber of Commerce (the "ICC"), and shall be finally settled under the Arbitration Rules of the ICC. The arbitration shall be conducted before a panel of three arbitrators. The complainant and the respondent to such dispute shall each select one arbitrator within thirty (30) days after giving or receiving the demand for arbitration. Such arbitrators shall be freely selected, and the parties shall not be limited in their selection to any prescribed list. Such two arbitrators shall select the third arbitrator. If either party to the arbitration does not appoint an arbitrator who has consented to participate within thirty (30) days after selection of the first arbitrator, the relevant appointment shall be made in accordance with the rules of the ICC. The place and location of the arbitration shall be London, England. The language to be used in the arbitral proceeding shall be English. The arbitrators shall be bound to the strict interpretation and observation of the terms of this Agreement and shall be specifically empowered to grant injunctions and/or specific performance and to allocate between the parties the costs of arbitration, as well as reasonable attorneys' fees and costs, in such equitable manner as the arbitrator may determine. The arbitration tribunal shall apply the Arbitration Rules of the ICC in effect at the time of the arbitration. However, if such rules are in conflict with the provisions of this Section 11.3, including the provisions concerning the appointment of arbitrators, the provisions of this Section 11.3 shall prevail. The arbitrators shall decide any dispute submitted by the parties to the arbitration strictly in accordance with the substantive law of the State of New York and shall not apply any other substantive law. Each party hereto shall cooperate with any party to the dispute in making full disclosure of and providing complete access to all information and documents requested by such party in connection with such arbitration proceedings, subject only to any confidentiality obligations binding on the party receiving the request. Judgment upon the award so rendered may be entered in any court having jurisdiction or application may be made to such court for judicial acceptance of any award and an order of enforcement, as the case may be. In no event shall a demand for arbitration be made after the date when institution of a legal or equitable proceeding based upon such claim, dispute or other matter in question would be barred by the applicable statute of limitations. Notwithstanding any of the foregoing, either party shall have the right, without waiving any right or remedy available to such party under this Agreement or otherwise, to seek and obtain from any court of competent jurisdiction any interim or provisional relief that is necessary or desirable to protect the rights or property of such party.
- 11.4 Injunctive Relief. No provision herein shall be construed as precluding a Party from bringing an action for injunctive relief or other equitable relief prior to the initiation or completion of the above procedure.

- **Relationship of the Parties.** Nothing in this Agreement is intended or shall be deemed, for financial, tax, legal or other purposes, to constitute a partnership, agency, joint venture or employer-employee relationship between the Parties.
- 12.2 Assignment
 - (a) Assignment by Licensee. Licensee may not assign this Agreement, in whole or in part, to any Affiliate or Third Party without the consent of SONNET. For clarity, a Sale is not deemed an assignment in the meaning of this Section 12.2(a).

-32-

- **(b)** No assignment under this Section 12.2 shall relieve the assigning Party of any of its responsibilities or obligations hereunder and provided, further, that as a condition of such assignment, the assignee shall agree to be bound by all obligations of the assigning Party hereunder.
 - (c) This Agreement shall be binding upon the successors and permitted assigns of the Parties.
 - (d) Any assignment not in accordance with this Section 12.2 shall be null and void.
- 12.3 Further Actions. Each Party agrees to execute, acknowledge and deliver such further instruments and to do all such other acts as may be necessary or appropriate in order to carry out the purposes and intent of this Agreement.
- 12.4 Rights Upon Bankruptcy. All rights and licenses granted under or pursuant to this Agreement are, and shall otherwise be deemed to be, for purposes of Section 365(n) of Title 11 of the United States Code and other similar laws in any jurisdiction outside the U.S. (collectively, the "Bankruptcy Laws"), licenses of rights to be "intellectual property" as defined under the Bankruptcy Laws. If a case is commenced during the Term by or against a Party under Bankruptcy Laws then, unless and until this Agreement is rejected as provided in such Bankruptcy Laws, such Party (in any capacity, including debtor-in-possession) and its successors and assigns (including a trustee) shall perform all of the obligations provided in this Agreement to be performed by such Party. If a case is commenced during the Term by or against a Party under the Bankruptcy Laws, and this Agreement is rejected as provided in the Bankruptcy Laws and the other Party elects to retain its rights hereunder as provided in the Bankruptcy Laws, then the Party subject to such case under the Bankruptcy Laws (in any capacity, including debtor-in-possession) and its successors and assigns (including a Title 11 trustee), shall provide to the other Party copies of all Information necessary for such other Party to prosecute, maintain and enjoy its rights under the terms of this Agreement promptly upon such other Party's written request therefor. All rights, powers and remedies of the non-bankrupt Party as provided herein are in addition to and not in substitution for any and all other rights, powers and remedies now or hereafter existing at law or in equity (including, without limitation, the Bankruptcy Laws) in the event of the commencement of a case by or against a Party under the Bankruptcy Laws.
- 12.5 Accounting Procedures. Each Party shall calculate all amounts hereunder and perform other accounting procedures required hereunder and applicable to it in accordance with either, as applicable (a) United States generally accepted accounting principles (US GAAP) or (b) International Financial Reporting Standard (IFRS), whichever is normally used by such Party to calculate its financial position, and in each case consistently applied by such Party.

-33-

- 12.6 Force Majeure. Neither Party shall be liable to the other Party or be deemed to have breached or defaulted under this Agreement for failure or delay in the performance of any of its obligations under this Agreement for the time and to the extent such failure or delay is caused by or results from acts of God, earthquake, riot, civil commotion, terrorism, war, strikes or other labor disputes, fire, flood, pandemic, failure or delay of transportation, omissions or delays in acting by a Governmental Body, acts of a government or an agency thereof or judicial orders or decrees or restrictions or any other reason which is beyond the control of the respective Party. The Party affected by force majeure shall provide the other Party with full particulars thereof as soon as it becomes aware of the same (including its best estimate of the likely extent and duration of the interference with its activities), and will use Commercially Reasonable Efforts to overcome the difficulties created thereby and to resume performance of its obligations hereunder as soon as practicable.
- 12.7 No Trademark Rights. No right, express or implied, is granted by this Agreement to a Party to use in any manner the name or any other trade name or trademark of the other Party in connection with the performance of this Agreement or otherwise.
- 12.8 No Sale for Resale. Upon notice from Licensee, SONNET shall make all reasonable best efforts to prevent the sale or distribution of the Product by any entity other than Licensee inside the Exclusive Territory (and if applicable, the Expansion Territory) and will not knowingly sell Product to anyone outside the Exclusive Territory (and if applicable, the Expansion Territory). Neither Licensee nor any Sublicensee of Licensee will knowingly sell, market, promote, distribute, co-market or co-promote or license rights to any Compound or Product to anyone outside the Exclusive Territory (and if applicable, the Expansion Territory) or in the Exclusive Territory (and if applicable, the Expansion Territory) for subsequent distribution or resale outside the Exclusive Territory (and if applicable, the Expansion Territory) and Licensee will take all reasonable precautions to prevent such license, distribution or resale outside the Exclusive Territory (and if applicable, the Expansion Territory).
- 12.9 Conflicting Rights. Neither Party will grant any right to any third party which would violate the terms of or conflict with the rights granted by such party to the other party pursuant to this Agreement.
- 12.10 Entire Agreement of the Parties; Amendments. This Agreement and the Schedules and Exhibits hereto constitute and contain the entire understanding and agreement of the Parties with respect to the subject matter hereof and cancel and supersede any and all prior negotiations, correspondence, understandings and agreements between the Parties, whether oral or written, regarding such subject matter except that the Mutual Confidentiality Agreement between the Parties dated [***] shall remain in full force and effect. No waiver, modification or amendment of any provision of this Agreement shall be valid or effective unless made in a writing referencing this Agreement and signed by a duly authorized officer of each Party.
- 12.11 Captions. The captions to this Agreement are for convenience only, and are to be of no force or effect in construing or interpreting any of the provisions of this Agreement.

-34

- 12.12 Governing Law. This Agreement shall be governed by and interpreted in accordance with the substantive English laws of the United Kingdom, excluding application of any conflict of laws principles that would require application of the Laws of a jurisdiction outside of the United Kingdom. Each Party irrevocably consent to the jurisdiction of London, United Kingdom.
- 12.13 Notices and Deliveries. Any notice, request, approval or consent required or permitted to be given under this Agreement shall be in writing and shall be deemed to have been sufficiently given if delivered in person, transmitted by facsimile (receipt verified) or by express courier service (signature required) to the Party to which it is directed at its address or facsimile number shown below or such other address or facsimile number as such Party shall have last given by notice to the other Party.

	If to SONNET, addressed to:	Stephen J. McAndrew, Ph.D. Vice President, Business Development 100 Overlook Center, Suite 102 Princeton, New Jersey 08540				
	If to Licensee. addressed to:	Rakesh K. Aggarwal, CEO 239 Arcadia Road #03-03, The Arcadia Singapore 289845				
	With a copy to:	Rachna Kairon, M.D., CTO Lot 203 & 205, Block D Kelana Square, 17, Jalan SS7/26 Kelana Jaya, 47310 Petaling Jaya Selangor Darul Ehsan, Malaysia				
12.14	condition for the future, or of an		v instance shall not be deemed or construed to be a waiver of such term or rtakings, obligations and agreements contained in this Agreement shall be ligation or agreement of either Party.			
12.15	of this Agreement is held to be p without invalidating the remainder	prohibited by or invalid under applicable Law, such provisi	on mer as to be effective and valid under applicable Law, but if any provision on will be ineffective only to the extent of such prohibition or invalidity ort to replace the invalid or unenforceable provision with a valid one which			
12.16		nay be executed in one or more counterparts, each of which acsimile copy of this Agreement, including the signature pag	will be deemed an original, and all of which together will be deemed to be es, will be deemed an original.			
		-35-				
NEW	nd binding effect as of the date first LIFE THERAPEUTICS PTE LTI / Rakesh K. Aggarwal	above written.	BIOTHERAPEUTICS, INC.			
	akesh K. Aggarwal		Mohan, Ph.D.			
C	hief Executive Officer	Chief	Executive Officer & Founder			
By: /s	/ Rachna Kairon	By: /s/ Step	ohen J. McAndrew			
	achna Kairon, M.D. hief Technology Officer		n J. McAndrew, Ph.D. resident, Business Development			
		-36-				
		SCHEDULE 1.12				
F###7		AMINO ACID SEQUENCE OF INTER	RLEUKIN-6			
[***]		Schedule 1.12-1				
		Schedule 1.12-1				
		SCHEDULE 1.19				
		EXISTING THIRD PARTY AGREE	EMENTS			
1.	License Agreement between Re	lief Therapeutics SA and ARES Trading SA (subsidiary o	of Merck KGaA) dated as of August 28, 2015.			
		Schedule 1.19-1				
		2 CANDAN D. 4 45				
		SCHEDULE 1.47 SONNET PATENTS				
[***]		SUNNET PATENTS				
LJ		Schedule 1.47				
		Solidare IIII				
		SCHEDULE 3.3				
	IL6 PRODUCT DEVELOPMENT TIMELINE					

CERTIFICATION OF THE CHIEF EXECUTIVE OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Pankaj Mohan, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q for the period ended March 31, 2021 of Sonnet BioTherapeutics Holdings, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 17, 2021 /s/ Pankaj Mohan

Pankaj Mohan Chief Executive Officer (Principal Executive Officer)

CERTIFICATION OF THE CHIEF FINANCIAL OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Jay Cross, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q for the period ended March 31, 2021 of Sonnet BioTherapeutics Holdings, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material
 information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in
 which this report is being prepared;
 - designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 17, 2021 /s/ Jay Cross

Jay Cross Chief Financial Officer (Principal Financial Officer)

CERTIFICATION OF CHIEF EXECUTIVE OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

I, Pankaj Mohan, certify that:

- 1. I am the Chief Executive Officer of Sonnet BioTherapeutics Holdings, Inc. (the "Issuer").
- 2. Attached to this certification is the Quarterly Report on Form 10-Q for the quarter ended March 31, 2021 (the "Report") filed by the Issuer with the Securities Exchange Commission pursuant to Section 13(a) or 15(d) of the Securities and Exchange Act of 1934, as amended (the "Exchange Act"), which contains financial statements.
- 3. I hereby certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to my knowledge:
 - The Report fully complies with the requirements of Section 13(a) or 15(d) of the Exchange Act; and
 - The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Issuer.

May 17, 2021

/s/ Pankaj Mohan
Pankaj Mohan
Chief Executive Officer
(Principal Executive Officer)

CERTIFICATION OF CHIEF FINANCIAL OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

I, Jay Cross, certify that:

- 1. I am the Chief Financial Officer of Sonnet BioTherapeutics Holdings, Inc. (the "Issuer").
- Attached to this certification is the Quarterly Report on Form 10-Q for the quarter year ended March 31, 2021 (the "Report") filed by the Issuer with the Securities
 Exchange Commission pursuant to Section 13(a) or 15(d) of the Securities and Exchange Act of 1934, as amended (the "Exchange Act"), which contains financial
 statements.
- 3. I hereby certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to my knowledge:
 - The Report fully complies with the requirements of Section 13(a) or 15(d) of the Exchange Act; and
 - The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Issuer.

May 17, 2021

/s/ Jay Cross Jay Cross Chief Financial Officer (Principal Financial Officer)